





How is the Minimum Wage Shaping the Wage Distribution: Bite, Spillovers, and Wage Inequality Carlos Oliveira

Gabinete de Estratégia e Estudos do Ministério da Economia e Mar Office for Strategy and Studies of the Ministry of Economy and Maritime Affairs Avenida da República, n.º 79, 1069-218 Lisboa – Portugal <u>www.gee.gov.pt</u> IOSN (online): 1647-6212





How is the Minimum Wage Shaping the Wage Distribution: Bite, Spillovers, and Wage Inequality

Carlos Oliveira^{a 1}

^aNova School of Business and Economics, Lisbon, Portugal

Abstract

Over the last three decades, wage inequality and the importance of the minimum wage presented an interesting negative correlation in Portugal. Using a semiparametric approach, counterfactual decomposition methods, and an extremely rich matched employer-employee dataset of all employees in the country, this paper presents significant visual and quantitative evidence of how the minimum wage structurally reshaped the wage distribution. The remarkable rise in the real minimum wage of 2006-2019 fully explained the sharp decline in wage inequality, and 40% of average wage growth - for women, who benefited the most, that was 60%. Spillover effects reached up to 40% above the minimum, being at times more important than the bite itself. The minimum wage reduced within and between wage inequality in several fronts, cutting the gender wage gap by a quarter, potentially decreasing the returns to education, and raising wages of workers at less productive firms. While the minimum wage bite was felt in workers' base wages, spillovers predominantly manifested in total wages.

JEL Classification: C14, D31, J31, J38

Keywords: minimum wage, spillover effects, wage inequality, counterfactual decomposition

Note: *This article is sole responsibility of the authors and do not necessarily reflect the positions of GEE or the Portuguese Ministry of Economy and Maritime Affairs*. Nota: Este artigo é da responsabilidade exclusiva dos autores e não reflete necessariamente as posições do GEE ou do Ministério da Economia e Mar.

Email address: carlos.oliveira@novasbe.pt (Carlos Oliveira)

¹I am extremely grateful to Pedro Portugal, who supervised the master's thesis that gave origin to this paper (Oliveira, 2021), providing vital guidance and assistance. The author gratefully acknowledges financial support from Funda ção para a Ciência e Tecnologia, Portugal under grant PTDC/EGE-ECO/7493/2020.





1. Introduction

Mounting wage inequality starting around the 1980s was a phenomenon experienced by most advanced economies (Milanovic, 2016; Hoffmann et al., 2020). It was primarily drivenby labour supply and demand dynamics, like skill-biased technological change (Card and DiNardo, 2002), globalisation (Feenstra and Hanson, 1996), and the general polarisation of the labour market (Autor et al., 2008), but the decay of labour market institutions in several countries, like the minimum wage and unionisation, has also been implicated (DiNardo et al., 1996; Koeniger et al., 2007).

Over a period where inequality in the upper half of the wage distribution follows a clear, positive trend across developed economies, the behaviour of inequality in the lower half of the distribution was much more diverse. Over the same period, the minimum wage, a policy that specifically targets the lower end of the distribution, also exhibited large cross-country heterogeneity, frequently in negative correlation with lower tail wage inequality. For instance, the real value of the federal minimum wage declined continuously in the United States, during the 1980s, and wage inequality rose dramatically (DiNardo et al., 1996; Lee, 1999), while the UK (Stewart, 2012) and Germany (Bossler and Schank, 2020) both saw wage inequality declining at the bottom of the distribution following the introduction of national minimum wages in 1999 and 2015, respectively. That is good reason to give it particular attention in research of inequality and the shaping of the wage distribution.

In Portugal, wage inequality was high and rising between the 1980s and the mid-2000s, and that increase was mostly due to a widening of the wage distribution at the top (Cardoso, 1998; Alvaredo, 2009; Centeno and Novo, 2014). After that, inequality actually decreased through a compression of the lower half of the wage distribution. In fact, the behaviour of wage inequality in Portugal since the mid-1980s can be well divided into three very distinct periods: between 1986 and 1994 inequality increased sharply, especially in the middle and upper-tail of the distribution; between 1994 and the mid-2000s changes in wage inequality were positive but small; and since the mid-2000s it decreased significantly, with an emphasis on the lower-half of the distribution. This is clearly illustrated in figure 1 and quantified in table C.6 of the appendix.





Figure 1: Measures of wage inequality. Panel A depicts wage inequality measures for the entire distribution. Panel B depicts measures of inequality at different parts of the distribution. Both standard deviation and the wage percentile differentials use the logarithm of total monthly wages. The two dashed vertical lines, in 1994 and 2006, divide the three periods of analysis.

The relatively short economic literature aiming to explain the changes in the Portuguese wage distribution over the last decades has naturally focused on the labour market dynamics side of the equation, which does explain a substantial part of those changes. Skill-biased technological changes and the overall modernisation of the labour market had a major impact in the widening of the wage distribution since the 1980s (Cardoso, 1997, 1998; Centeno and Novo, 2014), especially in what regards the growing disparity in returns to education throughout the wage distribution (Cardoso, 1998; Martins and Pereira, 2004; Machado and Mata, 2005) together with a remarkable increase in the education levels of Portuguese workers (Machado and Mata, 2005; Pereira, 2020), and also the rising share of women in the workforce (Cardoso, 1999; Pereira, 2020).

Still, after 1994 the widening of the wage distribution moderated and literature is yet to fully explain that phenomenon, aside from a possible excess of supply of skilled workers relative to unskilled ones that narrowed the lower tail of the distribution (Centeno and Novo, 2014), and a fading association between high-paying firms and high-wage workers (Portugalet al., 2018). Furthermore, wage inequality plummeted since the mid-2000s, especially for lower-wage workers - over a period where the nominal minimum wage almost doubled andthe share of workers on a base wage equal or lower than the minimum went from 8% up to 23% - and that remarkable phenomenon is yet to be studied.

With that in mind, this paper contributes to the literature by investigating a side of thestory acknowledged in no more than a handful of articles: the role of the minimum wage. Applying a semiparametric approach that bridges the distribution regression (Chernozhukov et al., 2013) and the rank regression framework (Fortin and Lemieux, 1998), explored in Fortin et al. (2021), and exploiting the extremely rich longitudinal administrative data of the individual records of *Quadros de Pessoal*, this paper presents estimates of the Portuguese wage density over three different time periods - 1986-1994, 1994-2006 and 2006-2019 - as well as a series of counterfactual scenarios that allow us to decompose changes in the distribution into the underlying change and the change attributed to the variation in the minimum. The effect of the minimum wage can then itself be decomposed into minimum wage "bite" effects and spillover effects. More, these results are not only presented in a visually clear way, but they are also quantified in several statistics of interest.

The results unambiguously show that the minimum wage was a crucial factor in theshaping of the Portuguese wage distribution when its relative importance increased. That is to say, while in the first period of analysis the minimum wage barely changed and its impacton the distribution was negligible, in the last period, a hike of 30% in its real value over 13years was enough to structurally reshape the wage distribution, which would





be much widertoday had the minimum not risen. For instance, the 90-10 percentile differential decreased by 22 log points between 2006 and 2019, and all of that decline can be attributed to theincrease in the minimum wage, while the 50-10 differential would have actually increased by 9 points instead of decreasing by 13. Between 2006 and 2019 the rise in the minimum wage was responsible for almost 40% of the average real wage growth of Portuguese workers. For workers at the lowest percentiles of the distribution, which were the ones who saw theirwages increase the most out of everyone in the workforce in reality, wages were stagnant in the counterfactual. The results also suggest that, in many cases, spillovers effects, which reached up to 40% above the minimum, can be even higher than the minimum wage "bite".

Breaking down the results by various segments of the population also demonstrated that: the minimum wage was most important for women sharply reducing inequality between females, being responsible for 60% of female average wage growth, and cutting the gender wage gap by 5 p.p., or a quarter; its impact was greater for less educated workers - explaining all of the wage growth of the least educated, and showing signs of a mitigating effect in the returns to education; it benefited more younger and older workers approaching retirement age; and was most influential for workers working at micro and large firms, and at less productive firms, demonstrating a potential productivity enhancing effect.

Finally, this paper uncovers a large discrepancy between the behaviour of base and total wages, which reacted differently to rises in the minimum wage - while the minimum wagebite was felt both in base and total wage, spillover effects were almost entirely reflected in total wages. This phenomenon gains importance knowing that both public debate and academic research frequently focus solely on base wages, overlooking a share of the minimum wage impact that this paper suggests can be as great as the impact of the bite itself.

The remaining of this paper is organised as follows: section 2 presents a review of the literature and relevant information on the minimum wage and wage inequality, section 3 presents the methodology and details on data and estimation, section 4 presents the visual and quantitative results of the paper along with a breakdown of the impact of the minimum wage on other wages, and section 5 concludes the paper with some final remarks.

2. The minimum wage and the wage distribution

The minimum wage allows for a set of margins through which labour markets can adjust, like employment, hours worked, wages and prices, but its impact across the wage distribution stands out in the sense of how structural it can be, depending on its value and the latent distribution.

When analysing the distributional effects of an increase in the real value of the minimum wage on wages – or the introduction of a minimum wage, for that matter – we must



consider three main ideas. First, the minimum wage can destroy jobs, leading to reduced incomes for those who would have been able to find a job in the absence of it. Second, and most notably, it allows some people at the bottom of the wage distribution to receive a higher wage. Third, the rise in wages can spillover to workers higher up in the pay-scale who wouldnot apparently benefit from the rise. There are many other mechanisms through which a minimum wage can affect wages, but their effects are either still unsettled in the literature or just not structural, by affecting only specific segments of the workforce and not the wage distribution *per se*.²

The first concept regards disemployment effects, which ought to raise *income* inequality since the minimum wage can be discarding the left tail of the wage distribution and cutting the incomes of all those workers that cannot find a job with the higher minimum (Stigler, 1946; Mincer, 1976; Neumark et al., 2004). The impact on *wage* inequality, however, will be the exact opposite, as the workers that are "discarded from the distribution" are those at its very end, ironically flattening the distribution (Machin et al., 2003; Teulings, 2003). Still, the empirical literature on these disemployment effects is extensive, and has indicatedthat job losses attributed to the minimum wage are oftentimes negligible (Brown et al., 1982; Machin and Manning, 1994; Card and Krueger, 1994, 1995; Dolado et al., 1996; Dubeet al., 2010). In fact, a suitably chosen minimum wage may even raise wages and employment simultaneously in a labour market with monopsony power (Katz and Krueger, 1992; Rebitzerand Taylor, 1995; Manning, 2003; Portugal and Cardoso, 2006).

While distributional outcomes of disemployment effects can almost be seen as unintended consequences, the two other mechanisms through which the minimum wage can shape the wage structure are distributional by nature. The minimum wage "bite" (Meyer and Wise, 1983a,b; Machin et al., 2003) is the most fundamental, and the most obvious: setting a minimum wage – or raising it – will increase the wages of those workers initially makingless than the minimum to exactly the level of the minimum, pushing those workers' wages at the left-most part of the distribution to the right, closer to the median, and compressingthe wage distribution as a whole. This bite is in many ways the primary motivation for a minimum wage (Freeman, 1996), and it is regarded as the main mechanism though which the minimum wage can affect wage dispersion.

Then there are the less evident spillover effects. Gramlich (1976) first suggested that pay raises could be spilling over to workers earning above the new minimum (Brown, 1999), but spillovers have been widely recognised in the literature since then (Grossman, 1983; Katz and Krueger, 1992; Lee, 1999; Cengiz et al., 2019; Fortin et al., 2021). The idea that a minimum wage can raise wages above that minimum came first from the notion of substitution effects, as the higher relative price of low-skilled workers would lead employers to substitute towards higher-skilled labour (Gramlich, 1976), although it shifted with time

² For instance, Freeman (1996) suggests that increasing the lower wages in a firm increases production costs, likely reducing either firms' shareholder income or higher-wage workers' wages. Firms may also have to increase prices, making those workers' wages, if not nominally, at least in real terms decreasing with the minimum wage.





to more of a relative wage concern, as firms mitigate the deterioration in workers' wages relative to the lower-skilled ones (Grossman, 1983).³ Spillover effects are also the hardest

to detect, but we at least know that their impact varies greatly, and that it is much higher for wages closer to the minimum.

Minimum wage research gained a lot of traction during the credibility revolution in labour economics (Angrist and Pischke, 2010; Belman and Wolfson, 2014), resulting in a wealth of empirical literature estimating the distributional effects of the minimum wage. The most influential papers study the United States over the last two/three decades of the 20th century, exploiting "the longest sustained decline in the real value of the federal minimum wage" inthe US along with a "dramatic rise in earnings and wage inequality" (Lee, 1999). The large majority come to similar conclusions: the falling real minimum wage accounted for much ofthe rise in dispersion in the lower tail of the wage distribution, without a significant effect on the upper tail of the distribution, particularly for women (Katz and Krueger, 1992; Card and Krueger, 1994; DiNardo et al., 1996; Lee, 1999). The higher wages of workers that would earn below the minimum - bite effects - are in all cases the most important driverof this compressing effect on the distribution, although spillovers often have an additional attenuating impact on inequality when they are considered (Lee, 1999; Teulings, 2003), while disemployment effects are rarely accounted for in distributional analyses, as their effect is assumed to be relatively unimportant (DiNardo et al., 1996).

Modern studies, with updated methodologies, keep confirming the inequality-reducing effects of the minimum wage (Card and Krueger, 2015; Autor et al., 2016; Cengiz et al., 2019; Fortin et al., 2021), although some stress that spillovers seem to have lost importanceover time, and studies outside the US have reached analogous conclusions.⁴

2.1. The Portuguese minimum wage

In Portugal, unlike many countries, the minimum wage rate is legislated at the national level. That monthly rate has been set by the government each January 1st almost every yearsince it was first legislated in 1974, and it covers nearly all workers in the country - the few exceptions, like the youth and the agricultural minimum wages, tended to disappear over time.

Sector-specific and firm-specific wage floors, through extended collective bargaining agreements, are very widespread in the Portuguese labour market (Cardoso and Portugal, 2005; Addison et al., 2017; Martins, 2021), although they are oftentimes renegotiated in line with the national minimum wage. In fact, the spillover effects of the minimum wage are regularly formally reflected in these negotiations, as employers are forced to negotiate

³ Spillovers can also arise in the presence of monopsony power, as unproductive firms close down when the minimum wage rises and their employees move to more productive, better-paying firms (Butcher et al., 2012; Dustmann et al., 2022).

⁴ See Machin (1997); Machin et al. (2003) for the UK, Bossler and Schank (2020) for Germany, Aeberhardt et al. (2016) for France, and Koeniger et al. (2007); Joe and Moon (2020) for OECD cross country analyses.



the wages higher up in the distribution in order to preserve job title wage premia.

The real value of the minimum wage grew steadily, but at a slow pace between the 1980s and the mid-2000s – only 10% in 20 years, between 1986 and 2006. After that, it grew ata much faster pace – 30% between 2006 and 2019 – with a slight slump between 2011 and 2014, when its nominal rate was frozen by the government in response to the financial crisis. The minimum wage relative to the average wage saw a severe drop between 1986 and 1994, decreased even more until 2006, although at a much lower pace, but started to increase dramatically since then. The share of workers earning the minimum wage or less decreased consistently but slowly until the mid-2000s but also increased rapidly since then, if we lookat the base wage - strikingly, this sudden rise is not reflected in the aggregate of total wages. To some extent, the importance of the minimum wage in the Portuguese labour market over time can be summarised in the same three periods that characterise wage inequality: it was decreasing until 1994, somewhat stable until 2006, and increased significantly since then. This is illustrated in figure 2 and quantified in table C.6 of the appendix.

So, if we contrast this behaviour with the behaviour of wage inequality over the years, especially in the lower tail, there is a noticeable symmetry that presents clear potential for research of minimum wage's distributional effects. However, the amount of research assessing the role of the minimum wage on wage inequality in Portugal is very limited, mostly concluding that it has had some narrowing effect on the wage distribution (Centeno



Figure 2: Importance of the minimum wage. Panel A depicts the logarithms of the real value of the minimum wage and the minimum wage relative to the average wage. Panel B depicts the share of workers with wages equal or lower than the minimum, using base and total wages (in 2002 and 2014 wage rounding near the minimum created noisy behaviour in the graph, so those years were omitted). The two dashed vertical lines, in 1994 and 2006, divide the three periods of analysis.

and Novo, 2014; Campos Lima et al., 2021), especially at the lower-tail (Carneiro et al., 2011) - only Pereira and Galego (2019) find no significant contribution of the minimum wage in attenuating wage inequality. Carneiro et al. (2011) acknowledge the presence of spillover effects, which they suggest had a limited impact on the distribution, although they do not get into that. There is also little literature on the disemployment effect in





Portugal, with unsettled, but generally slim impacts on employment and worker flows (Pereira, 2003; Portugal and Cardoso, 2006; Alexandre et al., 2021).

3. Methodology and data

To assess the effects of the minimum wage on wage inequality, the whole wage distribution is modelled by employing a framework that bridges distribution regressions and rank regressions, introduced by Fortin et al. (2021).⁵ After modelling the conditional wage distribution, we will be able to construct counterfactual scenarios by changing the distribution of certain covariates and reweighting the sample. This methodology will allow for the detailed assessment of the distributional effects of the minimum wage and its spillovers, assuming no disemployment effects, which is a relatively reasonable assumption in this context, according to recent literature.

3.1. Bridging distribution regressions and rank regressions

The distribution regression framework (Chernozhukov et al., 2013) can be used to estimate the probability of an outcome variable being below or above a given cutoff pointin a distribution. For instance, we can estimate the probability of the outcome variable *Y* being above the cutoff y_k as a flexible function of covariates *X* using a probit model as Prob($Y \ge y_k$) = $\Phi(X\beta_k)$.

Then we can model the whole distribution by dividing it in K - 1 bins, delimited by the cutoff points y_k and y_{k+1} - which may be chosen as percentiles of the unconditional distribution or using a fine grid - and computing the probability of the outcome variablelying inside each bin as:

$$\operatorname{Prob}(y_k \le Y < y_{k+1}) = \operatorname{Prob}(Y \ge y_k) - \operatorname{Prob}(Y \ge y_{k+1})$$
$$= \Phi(X\beta_k) - \Phi(X\beta_{k+1}) \quad \text{for } k = 1, 2, \cdots, K.$$
(1)

From these regressions, we can construct the conditional distribution from which all kinds of counterfactuals can be estimated, by changing the distributions of covariates or the regression coefficients themselves, allowing for the decomposition of changes in the wage distribution.

The general distribution regression framework as presented in Chernozhukov et al. (2013) is highly flexible, with no restrictions on how β_k coefficients vary across cutoffs, and that flexibility may be counterproductive when constructing the distribution, as we may get negative counterfactual probabilities and face identification problems in allowing for different

⁵ Fortin et al. (2021) examine the role of minimum wages spillover effects and unionisation threat effects in changes in the US wage distribution between 1979 and 2017, finding that spillovers magnify the explanatory power of decreasing minimum wages to two-thirds of the increase in inequality at the bottom of the female distribution.



 β_k at different points for all covariates, without restrictions.⁶ For that reason Fortin et al. (2021) propose a less flexible framework, closer to the rank regressions of Fortin and Lemieux (1998), where the effects of explanatory variables can be fixed or evolve smoothly across the distribution.

In Fortin and Lemieux (1998)'s rank regression framework the coefficients are fixed across the distribution. We can view it as a latent variable model where the latent variable index is given by $Y^* = X\beta + \varepsilon$, with $\varepsilon \sim N(0, 1)$, and the observed variable is the monotonic transformation $Y = g(X\beta + \varepsilon)$, with each observation's latent and observed variable rank being the same. Then we can divide the observed variable range by the cutoffs y_k and compute the probability of it being above a given cutoff as Prob $(Y \ge y_k) = \Phi(X\beta - c_k)$, where $c_k = g^{-1}(y_k)$ are the corresponding cutoffs in the latent distribution. This way, we can model the distribution through an ordered probit where each category is a bin delimited by y_k and y_{k+1} with probability:

$$Prob(y_k \le Y < y_{k+1}) = \Phi(X\beta - c_k) - \Phi(X\beta - c_{k+1}) \quad \text{for } k = 1, 2, \cdots, K.$$
(2)

Here β coefficients are fixed across the distribution, as opposed to the β_k coefficients in equation 1. But the flexibility can also be advantageous. That is why we will employ a framework that bridges both methods, by introducing heteroscedasticity. Incorporating the interaction between X and y_k in the model allows for the effect of explanatory variables to evolve linearly across the wage distribution, such that $Prob(Y \ge y_k) = \Phi(X\beta + y_kX\gamma - c_k)$, and therefore:

$$Prob(y_k \le Y < y_{k+1}) = \Phi(X\beta + y_k X\gamma - c_k) - \Phi(X\beta + y_{k+1} X\gamma - c_{k+1})$$
(3)

As Fortin et al. (2021) demonstrate, assuming the same effect of covariates across the distribution would be too strong of an assumption to make when modelling the distribution of wages. More so given the existing literature on how the effects of education, in particular, have been rather disparate throughout the distribution in the Portuguese labour market (Machado and Mata, 2005).

3.2. Minimum wage effects

From this kind of framework, minimum wage effects can be estimated straightforwardly, by simply adding a set of dummy variables that indicate the distance to the minimum. We know that the effect of a minimum wage on the cumulative wage distribution would be the generation of a spike over that point. For instance, if in the latent wage distribution without the minimum wage we had $Prob(Y \ge y_k) = \Phi(X\beta - c_k)$ (using the

 $^{^{6}}$ For instance, the final model will include a set of year dummies to capture macroeconomic conditions, which could absorb the whole effect of the minimum wage if we allowed for full flexibility of the β k coefficients.





rank regression model for simplification), if y_k is the cutoff point just above the minimum wage $(y_{k-1} \ge MW < y_k)$, the probability of Y being above that cutoff point in the observed distribution can be characterised as $\operatorname{Prob}(Y \ge y_k) = \Phi(X\beta - c_k + \delta_0)$, where δ_0 is a parameter that captures the effect of the minimum wage. In fact, $\operatorname{Prob}(Y \ge y_k) = \Phi(X\beta - c_k + \delta_0)$ for all y_k above the minimum wage.

Furthermore, in the presence of spillover effects, they can also be accounted for with a similar strategy. For example, if the effect of the minimum wage spills over to only one wage bin above the minimum wage, we have $\operatorname{Prob}(Y \ge y_k) = \Phi(X\beta - c_k + \delta_0 + \delta_1)$ for all $y_{k-1} > MV$.

With this in mind, we can estimate the effect of the minimum wage, above and below, by adding a set of dummy variables $D^{m}_{kt} = 1[yk-m \le MW_t]$ indicating the distance to that year's minimum wage, such that $Prob(Y \ge yk) = \Phi(X\beta + \Sigma_m D^m_{kt} \delta_m - c_k)$, or in our framework:

$$\operatorname{Prob}(Y \ge y_k) = \Phi\left(X\beta + y_k X\gamma + \sum_m D_{kt}^m \,\delta_m - c_k\right) \tag{4}$$

3.3. Empirical model and estimation

To model the entire wage distribution, the log wage range is divided into 60 wage bins between 4.5 and 7.5 (a range that comprises more than 99% of all observations), where eachbin has a length of 5 log points, plus the first and the last bin, for a total of 62 bins (see the corresponding histograms in figure 3). Then we estimate one stacked probit model, by stacking 62 copies of the original dataset, and adding the outcome variable, equal to 1 if $y_{it} > y_k$ and zero otherwise, for $k = 1, \dots, 61$ (this stacking is illustrated in table C.7 of the appendix). This way we can allow the coefficients to evolve linearly throughout the distribution. The full model is:

$$\operatorname{Prob}(Y_{it} \ge y_k) = \Phi\left(X_{it}\beta + y_k X_{it}\gamma + \sum_{m=b}^a D_{kt}^m \,\delta_m - c_k\right) \qquad \text{for } k = 1, 2, \cdots, 61.$$
(5)

The covariates X_{it} include the set of individual characteristics - gender, age and squared age, tenure and squared tenure and years of education - as well as 18 age-education interactiondummies in order to capture supply and demand dynamics (Bound and Johnson, 1992), year and region fixed effects. Some specifications also include a "heaping variable".⁷ Then we have the interaction term $y_k X_{it}$, although not all covariates are interacted with the cutoff points for computational convenience - only age-education

⁷ Autor et al. (2016), in their critique of some minimum wage empirical research methods, suggest that the heaping of

hourly wages at integer values in the US could be creating spurious spillovers when the minimum wage is slightly below an integer. Fortin et al. (2021) correct for that by including this type of variable which aims to capture the effect of heaping. Here an analogous strategy is used to capture heaping at multiples of $50 \in$.



interactions and year dummies are included here, capturing both supply and demand and education changes at different points of the distribution - which are the two main explanatory factors of changes in wage inequality according to the literature - as well as macroeconomic conditions. Minimum wage effects are captured in the δ_m coefficients, through the dummies D^m_{kt} , where $m \in \{b, a\}$ and (b < 0 < a). δ_0 measures the jump in probability at the minimum, $\delta_{m>0}$ measure the spillovers and $\delta_{m<0}$ measure any decrease in probability below the minimum. The number of bins above and below the minimum wage at which the effect is measured is different for different specifications, as these effects are not always the same. We will go further into this section 4, but these dummies were allowed for up to 3 bins below and 8 bins above the minimum wage bin. Finally, c_k are wage bin dummies.

3.4. Details on data

The analysis in this paper makes use of the incredibly rich longitudinal matched employeremployee dataset of *Quadros de Pessoal* (QP), an annual mandatory survey of all establishments in Portugal with at least one wage earner, collected by the Portuguese Ministry of Labour and Social Solidarity. Among a remarkably extensive list of variables on firms, establishments, and workers, the QP dataset provides demographic information such as age, gender and education, and employment information like base and total monthly wages, hours worked, tenure at the firm and occupation, for all employees in Portugal since 1982 (except for 1990 and 2001, when it was not available). Few matched employer-employee datasets are as detailed and precise as QP, since employers are not only required to provide their full roster of employees to the Ministry, but they must also ensure that workers are provided with that information, in order to monitor compliance with labour law provisions.

All estimation exploits a random sample of 10% of all full-time workers between the ages of 18 and 64, from 1986 to 2019 - between 111, 987 observations in 1986 and 226, 191 in 2019. The random 10% sample is drawn for computational convenience, in order to handle a stacked dataset of 62 x *N* observations, since there is no reason to believe that any problem would arise from the use of a random sample given the large size of the original dataset and the nature of the estimations. The variable of interest in most cases is the total monthly wage (deflated to 1986 CPI), which is most relevant in measuring wage inequality, and other variables utilised include age, gender, tenure, region and level of education (transformed into years of education). Still, base wages will also be utilised, since they presented a rather different behaviour compared to total wages, but this phenomenon will be further discussed in section 4.4 - see figure 3.





Figure 3: Distribution of wages from 1986 to 2019. This figure depicts the distribution of the logarithm of real wages across the years using total wages and base wages. Histograms have the same wage bins that areused in the estimation of the model (0.05 log bins, between 4.5 and 7.5). The black vertical line indicateseach year's minimum wage, while the dashed lines indicate the reference years' minimum wages as they appear. This is an interactive figure - click play or use the controls to see the distribution moving over the years (Acrobat Reader recommended).

4. Results

As discussed in section 2.1, the joint behaviour of wage inequality and the minimum wage between 1986 and 2019 can be organised into three distinct periods: 1986-1994, whereinequality grew and the importance of the minimum wage was stagnant, 1994-2006, whereinequality stabilised and the minimum wage rose modestly, and 2006-2019, where inequality significantly decreased and the importance of the minimum wage increased unprecedentedly.

For a most complete analysis, the model laid out in section 3.3 was separately estimated for the three time-periods. The coefficients of interest resulting from that estimation - the coefficients measuring minimum wage effects, δ_m - are reported in table 1, specifications (1-3). After looking at the results for the three periods, we will focus the analysis on the lastone, 2006-2019, which was the most interesting, in order to dissect even further the impactof the minimum wage on the wage structure. Discussion in section 3.4 suggested that base wages could be reacting to changes in the minimum wage in a different way than total wages, so a model was also estimated for the period 2006-2019 using base wages instead of total wages, reported in table 1, specification (4).





δ_m coefficients	1) 1986-1994	Total wages 2) 1994-2006	3) 2006-2019	Base wages 4) 2006-2019
10%-15% below MW (δ_{-3})	0.096	0.459	/	,
5%-10% below MW (δ_{-2})	0.127	0.370		
0%-5% below MW (δ_{-1})	0.045	0.336		
At the minimum wage (δ_0)	0.302	1.092	1.097	1.649
0%-5% above MW (δ_1)		0.068	0.117	0.166
5%-10% above MW (δ_2)		0.052	0.145	0.067
10%-15% above MW (δ_3)		0.067	0.172	()
15%-20% above MW (δ_4)		0.057	0.121	
20%-25% above MW (δ_5)		0.042	0.086	
25%-30% above MW (δ_6)		()	0.055	
30%-35% above MW (δ_7)			0.033	
35%-40% above MW (δ_8)			0.021 (0.001)	
Number of observations	34724200	71344988	99422358	70180488

Table 1: Estimated coefficients of minimum wage effects. This table reports the probit regression estimates for the δ_m coefficients in equation 5. All specifications include the explanatory variables mentioned in section 3.3, although the heaping variable is only present in specifications (3) and (4), where the heaping of wages was more obvious. Standard errors, in parentheses, are clustered by individual

The range of *m*, the maximum distance to the minimum at which minimum wage effects are allowed, above and below, was differently determined for each specification by excluding insignificant coefficients, in order to best fit the model to each case. For instance, while minimum wage effects are identified up to 40% above the minimum in the period 2006-2019, no spillovers are found in 1986-1994. As such, the adopted ranges of *m* were $\{-3, 0\}, \{-3, 5\}, \{0, 10\}$ and $\{0, 2\}$ for specifications (1), (2), (3) and (4), respectively.

These are probit regression coefficients so their direct interpretation can be problematic, but we already see that they are all positive, which is expected since they refer to the cumulative distribution. We can also identify the spike created by the minimum wage in the $\delta 0$ coefficients, which are much higher than other δ_m , and see that spillover effects tend to decrease as they draw away from the minimum, as coefficients are decreasing for m > 0.

Still, the beauty of this kind of methodology does not come from the coefficients themselves, but from its ability to let us play with all kinds of counterfactual scenarios in order to extract, from specific statistics, or from the entire distribution, the effect associated to certain explanatory factors. In this paper, a series of counterfactuals scenarios are constructed in order to assess the effects of the minimum wage at different points of the distribution – which is straightforward after the estimation of the conditional wage distribution.





We will first look at a series of actual and counterfactual density estimates of the wage distribution, which are obtained using an adaptive kernel density estimator (Abramson, 1982). The construction of counterfactual densities is done through the reweighting approach of DiNardo et al. (1996) whereby samples are appropriately weighted in a similar way to common propensity score methods (Fortin et al., 2011). Then, we will quantify the effect of the minimum wage in several statistics of interest, decomposing changes in those statistics into the underlying change, minimum wage bite effects and minimum wage spillover effects.

In short, counterfactual densities and statistics are obtained by changing the distribution of certain covariates - in this case, what changes is the distribution of the D^{m}_{kt} minimum wage effect dummy variables - and then reweighting the sample in accordance to the new distributions of the covariates. For instance, to compute the wage the distribution that would have prevailed in 2019, had the minimum wage stayed at its 2006 level, we simply compute the above-mentioned probabilities, using the parameters that we estimated, but switching the distribution of the minimum wage effect dummies with their 2006 level, $D^{m}_{kt=2006}$. If the objective is to estimate the 2019 counterfactual wage distribution in the absence of spillovers, then we simply set the minimum wage effect dummies above the minimum - the spillovers - to zero: $D^{m}_{kt}=2019 = 0$ for m > 0. After that, we can estimate counterfactual wage densities and distributional statistics by reweighting the sample through the use of the reweighting factor $\psi(X_k)$ (DiNardo et al., 1996), estimated as

$$\hat{\psi}(X_k) = \frac{\bar{Y}_k^C(X)}{\bar{Y}_k(X)}, \quad k = 1, \cdots, K$$
(6)

where \overline{Yk} are the averages of the outcome variable, per wage bin, in the actual scenario, and \overline{Yk}^{C} are the averages of the outcome variable in the counterfactual.⁸

4.1. Counterfactual wage densities

All in all, the wage distributions that would have prevailed in 1994, 2006 and 2019, hadthe minimum wage remained at its 1986, 1994 and 2006 level, respectively, are presented in figure 4, while the corresponding changes in density between the actual and the counterfactual can be seen in figure 5. Figure D.17 of the appendix portrays the counterfactual densities of the last year, if the minimum wage had risen but no spillovers occurred sincethe first year, to aid disentangling the role of spillover effects.

As we can see in figure 4, while the wage distribution visibly widened to the right between 1986 and 1994, the real value of the minimum wage in 1986 was nearly the same eight yearslater. Unsurprisingly, the mere 1% real increase in the minimum wage over this

⁸ The reweighted adaptive kernel density estimator, used in the estimation of the wage densities, is very briefly laid out in Appendix A.



period had nearly zero impact on the 1994 wage density, which is almost identical to the actual wage density, except for a slight smoothing around the minimum which can be attributed to a lack of common support⁹ The absence of spillovers, reflected in specification (1), makes the counterfactual in figure D.17 indistinguishable from the actual 1994 density, so we conclude that the unchanged minimum wage had no discernible effect on the wage density between 1986 and 1994.

In the second period of analysis we start to notice some minimum wage effects. Over twelve years, the minimum wage increased 9%, which is not much, but certainly enough to reshape the wage distribution to some extent. In figure 4 we can see that the mass of workers earning wages below the 2006 minimum would be much more representative hadthe minimum wage not risen by those 9%, and that a visible portion of the workers thatearned wages up to 25% above the 2006 minimum would be earning lower-wages, many of them earning less than that new minimum. Figures 5 and D.17 confirm that the rise in the minimum wage was not only responsible for a rise in wages for those who are now earning the new minimum, but it also spilled over to workers that would not be affected by a higher minimum at first sight.

However, it is between 2006 and 2019 that we can observe the true potential of a minimum wage. Looking at the actual wage densities of 2006 and 2019 in figure 4 we can see that the distribution shifted to the right over 13 years, but that mainly happens through a significant compression of the distribution at the bottom. And watching the wage histograms move throughout the years in the interactive figure 3 does hint that it is the minimum wage that is pushing that left-tail more and more to the right. But it is by looking at the 2019 counterfactual density with the 2006 minimum wage, in figure 4, that we can truly graspthe effect of the minimum wage on the shape of the distribution.

Had the real minimum wage not risen by the impressive 29% that it did, the shape of the distribution would be incredibly different in 2019, with the very high mass of workers earning wages close to the minimum, that we see in the actual distribution, significantly diluting, with a very large portion of workers shifting back in the pay-scale to wages much lower than the 2019 minimum, and many remaining at the 2006 minimum. And if the effect

⁹ This type of weighting method can always be subject to problems of common support (Frolich, 2004; Fortin et al., 2011), especially taking into account the binding nature of the minimum wage in the Portuguese context.











Figure 4: Counterfactual densities for the periods 1986-1994, 1994-2006, and 2006-2019. Each panel of this figure depicts the counterfactual density estimate for the last year, had the minimum wage stayed at its first year level, in blue, as well as the actual estimates for the first and the last year of each period of analysis, in red and green, respectively. The dashed vertical lines indicate the first and last years' minimum wages.







Figure 5: Changes in density attributed to the change in the minimum wage. This figure depicts the difference in densities between the actual last year distribution and the counterfactual distribution with first year's minimum wage for the three time periods. Dashed vertical lines indicate the minimum wages.

of spillovers is not clear enough, figures 5 and D.17 make it even more evident that a verylarge share of lower-wage workers saw their wages increase significantly, and that portiondid not "accumulate" at the new minimum wage, but it actually spilled-over quite abovethat, to wages up to 40% above the 2019 minimum.

4.2. Quantifying minimum wage effects

After recognising the clear visual impact that the minimum wage has had on the wage distribution, it is important to quantify that impact. The distributional effects of the changes in the minimum wage were quantified into several statistics of interest, laid out in table 2. The table is divided into three panels, for the three time-periods, where the columns present those statistics, measuring inequality across the distribution (standard deviation and 90-10 percentile differential), upper-tail and lower-tail inequality (90-50 and 50-10 differentials), the share of workers on the first and last years' minimum wages (incidence of the minimumwage), and the average wage. The first four rows of each panel present these statistics for the different scenarios that were estimated - first year, last year, counterfactual last year with first year's minimum wage, and counterfactual last year without spillovers. The three rows after that present the actual change in those statistics between the first and the last year, the *total change*, as well as the decomposition of that change into the *underlying change* - the change that would have occurred between those years if the real value of the minimumwage had stayed at its first year level - and the *change due to the minimum wage* - the change driven by the shift in the minimum wage.

The two final rows of each panel present the further decomposition of the *change due to the minimum wage* into the *change due to the minimum wage bite*, the change driven by the workers whose wages would be below the new minimum but the minimum wage managed to push up to that minimum, and the *change due to spillovers effects*, driven by





the workers who would got their pay pushed even higher than the new minimum.¹⁰

	63	SJ 00.10 00.50 50.1		EO 10	Incidence	Incidence	Mean
	50 50-10 50-50 50-10		01-06	1st year	last year	wage	
	A) 1986-1994						
1986	0.5	1.21	0.79	0.42	0.11	0.11	5.24
1994	0.57	1.41	0.92	0.49	0.02	0.06	5.42
1994 w/ 1986 MW	0.58	1.4	0.92	0.48	0.03	0.05	5.43
1994 w/o spillovers	0.57	1.41	0.92	0.49	0.02	0.06	5.42
Total change	0.07	0.2	0.13	0.07	-0.09	-0.05	0.18
Underlying change	0.08	0.19	0.13	0.06	-0.08	-0.06	0.19
Change due to MW	-0.01	0.01	0	0.01	-0.01	0.01	-0.01
Change due to bite	-0.01	0.01	0	0.01	-0.01	0.01	-0.01
Change due to spillovers	0	0	0	0	0	0	0
		B) 19	994-2006				
1994	0.57	1.41	0.92	0.49	0.06	0.11	5.42
2006	0.58	1.43	0.95	0.48	0	0.04	5.57
2006 w/ 1994 MW	0.6	1.47	0.95	0.52	0.05	0.07	5.56
2006 w/o spillovers	0.58	1.47	0.94	0.52	0	0.06	5.57
Total change	0.01	0.02	0.03	-0.01	-0.06	-0.07	0.15
Underlying change	0.03	0.06	0.03	0.03	-0.01	-0.04	0.14
Change due to MW	-0.02	-0.04	0	-0.04	-0.05	-0.03	0.01
Change due to bite	-0.02	0	-0.01	0	-0.05	-0.01	0.01
Change due to spillovers	0	-0.04	0.01	-0.04	0	-0.02	0
		C) 20	06-2019				
2006	0.58	1.43	0.95	0.48	0.04	0.23	5.57
2019	0.51	1.21	0.86	0.35	0	0.04	5.73
2019 w/ 2006 MW	0.57	1.43	0.86	0.57	0.03	0.16	5.67
2019 w/o spillovers	0.53	1.34	0.85	0.49	0	0.1	5.71
Total change	-0.07	-0.22	-0.09	-0.13	-0.04	-0.19	0.16
Underlying change	-0.01	0	-0.09	0.09	-0.01	-0.07	0.1
Change due to MW	-0.06	-0.22	0	-0.22	-0.03	-0.12	0.06
Change due to bite	-0.04	-0.09	-0.01	-0.08	-0.03	-0.06	0.04
Change due to spillovers	-0.02	-0.13	0.01	-0.14	0	-0.06	0.02

Table 2: Decomposition of changes in statistics of interest for the three periods. Each panel depicts statistics of interest for the first year, the last year, and the counterfactuals of the last year with the first year minimumwage and with no spillovers. They also contain the change in these statistics between the first and the last year, decomposed into the underlying change and the change attributed to the shift in the minimum wage, which is then itself decomposed into minimum wage bite effects and spillover effects.

The minimum wage practically stagnated during the first 8 years of the sample, sothe deviation across the different 1994 scenarios for all statistics is negligible. Inequality increased rapidly, especially at the upper tail, the average wage increased too, but virtually all of it is explained by the underlying changes in the economy and not by any minimum wage changes.

Between 1994 and 2006 there was a 9% increase in the minimum wage and the

 $^{^{10}}$ The decomposition of results is explained with more detail in Appendix B.



growthin inequality slowed. The underlying increase in the standard deviation of log wages wasonly 3 log points over 12 years, much lower than the 8 points in the previous 8 years, sothere clearly were other more important factors mitigating wage inequality. Nonetheless, the minimum wage was still able to cut the increase in the standard deviation and the 90-10 percentile differential by two thirds, and was actually responsible for the 1 point decrease in the 50-10, which increased by 3 points in the counterfactual. Notably, if the minimum wage had not gone up, 5 of the 6% of workers that were on the minimum wage in 1994 would still be earning that value in 2006.

Having said that, it is again in the third period that we see the full impact of the minimum wage. As discussed before, wage inequality decreased greatly between 2006 and 2019 - the standard deviation of log wages decreased by 7 points, the 90-50 differential decreased by 9 points and the 50-10 by 13, for a total decline of 22 points in the 90-10 differential. When we decompose those changes we find that, as the wage density estimates suggested, the minimum wage had a crucial role in those developments.

Had the minimum wage not increased, the standard deviation would have decreased by only one log point - the minimum wage bite was able to reduce the standard deviation by4 extra log points and its spillovers managed to reduce it another 2 points. More striking,the 90-10 differential would not have changed at all - while upper-tail inequality decreased by 9 log points due to other factors, lower-tail inequality would have increased by 9 points, offsetting any changes in the 90-10 differential. Instead, it decreased by 22 points, 13 of those in the 50-10, and most of that was due to spillovers, which alone reduced the differential by14 points.

Figure 6 perfectly illustrates this impact on wage inequality: while the actual 2006-2019 wage growth incidence curve was straightly decreasing - meaning that wage growth between those years was almost inversely proportional to a worker's position in the wage distribution, with wages at the very bottom of the distribution growing by 30%, while wages at the verytop almost stagnated - if the minimum wage had not gone up we would be seeing an reverse-U-shaped curve - whereby the wages of workers in the lowest percentiles of the distribution, the ones who saw their wages increase the most in reality, would have actually been theores with the smallest growth. In figure D.18 of the appendix, we can see these curves for the first two periods, where the difference between the actual and counterfactual curves is almost trivial compared to the third period.

In 2019, no worker was earning a wage equal or lower than the real value of the 2006 minimum wage. However, had the minimum wage not risen, the share of workers earning the 2006 minimum wage in 2019 would still be 3%. But what is more striking is the share of workers earning the the real value of the 2019 minimum in both years. In 2019 the minimum wage was \in 600, and its incidence was only 4%, but the share of workers earning that







Figure 6: Wage growth incidence curves, 2006-2019. This graph depicts the actual and counterfactual growth rate of wages by wage percentile from 2006 to 2019. The shaded horizontal lines indicate the mean wage growth. The first percentile was omitted. The slight difference between the actual and counterfactual curves at the higher percentiles of the distribution is due to the reweighing, not representing actual lower wage growth.

value in 2006, in real terms, was a staggering 23%. This 19 point decrease over 13 years is due in part to other changes in the economy, but mostly due to the increase in the minimum wage - the minimum wage bite put 6% of the workforce above that threshold and the spillovers put another 6%.

In the end, such a powerful impact of the minimum wage across the distribution could only translate itself in the average wage. And it did. The average wage increased by 16 points over 13 years. That is not much, but it would have been only 10 points if the minimum had not risen by those 29%. The rise in the minimum wage was responsible for almost 40% of the growth in the average wage between 2006 and 2019.

4.3. Breaking down minimum wage effects

The two previous subsections clearly documented the unequivocal impact of the minimum wage on the wage distribution. In general, but even more so in the third period - which is the most interesting, since it was where the minimum wage grew the most and its impact was greater - the rise in the minimum wage completely reshaped the wage distribution, triggered a sharp decline in wage inequality, and lifted the average wage up. This subsection will focus on that third period in order to dissect even further the impact of the minimum wage, by looking at differences within and between different segments of the workforce. We will look at differences between and within genders, education levels, between workers of different ages, and workers' firm characteristics.

4.3.1. Gender

The minimum wage is generally recognised as having a greater impact on women than



Gabinete de Estratégia e Estudos

men, since women tend to be over-represented among low-wage workers (Card et al., 2016) - in Portugal, 2019, women represented only 46% of the workforce, but 58% of minimum wage earners. To understand that different impact of the minimum wage within and between genders, the actual and counterfactual female and male wage densities were estimated. They are portrayed in figure 7.



Figure 7: Counterfactual densities by gender. This figure depicts the counterfactual density estimate for the last year, had the minimum wage stayed at its first year level, in blue, as well as the actual density estimates for the first and the last year of each period of analysis, in red and green, respectively, for female and male workers, analogously to figure 4.

We already saw that the 29% rise in minimum wage significantly compressed the wage distribution at the bottom, pushing a large mass of workers higher up in the pay-scale. When we estimate the female and male distributions separately, the impact of the rise in the minimum turns out to be similar in shape for both: a significant compression at the bottom, with a large share of workers receiving higher wages. However, the female wage distribution was already much more compressed at the minimum in 2006 - while the male distribution was wider and further to the right - so the impact was much more significant for women.

Table 3 reports analogous results to table 2, but divided by gender (for 2006-2019).





The quantitative results display even more clearly how the minimum wage was much more influential for women than for men. While male wage inequality would have declined over this period anyway - since the underlying change was negative for all inequality indicators

- female wage inequality would have actually increased, had the minimum wage not risen

- there would have been an increase of the standard deviation (2 points) and the 90-10 differential (7 points), pushed by the sharp increase in lower-half wage inequality (increase of 14 points in the 50-10 differential), even with the decrease in upper-half inequality (7 points). The actual change in within-gender inequality indicators was comparable for both genders, but the share of that change that can be attributed to the shift in minimum wage was much greater for women.

	e.a	a 00.10 00.50	50.10	Incidence	Incidence	Mean	
	bu	90-10	90-50	50-10	1st year	last year	wage
		A) Fer	nales ($\Delta \Lambda$	W = 0.29			
2006	0.52	1.26	0.88	0.38	0.05	0.32	5.44
2019	0.46	1.08	0.79	0.29	0	0.04	5.64
2019 w/ 2006 MW	0.54	1.33	0.81	0.52	0.05	0.24	5.52
2019 w/o spillovers	0.48	1.17	0.8	0.37	0	0.13	5.61
Total change	-0.06	-0.18	-0.09	-0.09	-0.05	-0.28	0.2
Underlying change	0.02	0.07	-0.07	0.14	0	-0.08	0.08
Change due to MW	-0.08	-0.25	-0.02	-0.23	-0.05	-0.2	0.12
		B) M	ales (ΔM)	W = 0.29)			
2006	0.6	1.48	0.95	0.53	0.03	0.16	5.68
2019	0.54	1.29	0.88	0.41	0	0.03	5.82
2019 w/ 2006 MW	0.57	1.39	0.87	0.52	0.02	0.08	5.81
2019 w/o spillovers	0.56	1.44	0.87	0.56	0	0.09	5.8
Total change	-0.06	-0.19	-0.07	-0.12	-0.03	-0.13	0.14
Underlying change	-0.03	-0.09	-0.08	-0.01	-0.01	-0.08	0.13
Change due to MW	-0.03	-0.1	0.01	-0.09	-0.02	-0.05	0.01

Table 3: Decomposition of changes in statistics of interest by gender, 2006-2019. This table depicts the statistics of interest, analogously to table 2, separately for females and males.

More, the impact of the rise of the minimum wage on the average wage was completely disparate for men and women: men's average wage grew by 14 log points between 2006 and 2019, and only 1 of those 14 points (7%) was due to the rise in the minimum wage, while women's average wage grew by 20 points and 12 of the 20 points (60%) can be attributed to that shift.

Such a large difference in the impact of the minimum wage between genders may seem surprising at first, but is almost expected if we are aware of the large differences between the female and male distributions. Figure 7 showed how the female distribution was much more compressed at the minimum beforehand, and if we look, for instance, at the share of workers earning the real value of the 2019 minimum wage or less in 2006, we can see that it was 32% of women and only 16% of men. This large difference is also



consistent with the international economic literature, as mentioned in section 2, since women tend to be over-represented in the low-wage worker population of most countries.

Finally, it would be important to understand if the minimum wage had an impact on the gender wage gap. Economic literature has suggested that the minimum wage can trigger a compression of the adjusted differential between male and female wages.¹¹ As Rubery and Grimshaw (2011) put it: "a minimum wage can act as a strategic instrument in countering distortionary effects and smoothing out the imbalances that result from the interactions between sex segmentation of labour supply and sex segregation in employment with processes of wage-setting".

To address that, a simple wage equation was estimated for 2006, for 2019 and for the two 2019 counterfactual scenarios, with the reweighted samples. The coefficients of the main explanatory variables are reported in table 4, while the covariates not included in the table were: age and squared age, tenure and squared tenure, region, industry, and professional category.

	1) 2006	2) 2019	3) 2019 w/ 2006 MW	4) 2019 w/o spillovers
male	0.199 (0.002)	0.167 (0.002)	0.216 (0.007)	0.173 (0.002)
education				
2nd cycle	0.088 (0.003)	0.058 (0.003)	0.070 (0.009)	0.056 (0.004)
3rd cycle	0.178 (0.003)	0.126 (0.003)	0.133 (0.013)	0.125 (0.004)
secondary	0.307 (0.003)	0.232 (0.003)	0.249 (0.010)	0.244 (0.004)
higher	0.636 (0.006)	0.488 (0.004)	0.532 (0.013)	0.518 (0.005)
Number of observations	207539	226187	226187	226187

Table 4: Estimated coefficients of interest from the log wage equation. The explanatory variables not included in the table were: age and squared age, tenure and squared tenure, region, industry, and professional category. Robust standard errors are reported in parentheses.

In 2006, a man would receive a 19.9% higher wage than a woman with the same observable characteristics. 13 years later, in 2019, that gap had decreased by 3.2 points, to 16.7%. However, had the minimum wage not risen, the gender wage gap would have actually increased to 21.6% in 2019. That means that the rising minimum wage was responsible for a decrease in the gender wage gap of 4.9 percentage points, almost a quarter of the total gap, which would have actually increased, had the minimum wage not risen. And almost all of that effect was due to the minimum wage bite (4.3 of the 4.9 points). This suggests that, when employers were forced to increase their employees' wages due to the rise in the minimum, they avoided paying the discrimination component

¹¹ See evidence from the UK (Robinson, 2002; Bargain et al., 2019), Germany (Caliendo and Wittbrodt, 2021), Poland (Majchrowska and Strawi 'nski, 2018), China (Li and Ma, 2015), Korea (Cho and Yang, 2021), or Canada (Shannon, 1996). Interestingly, one offbeat paper on Portuguese young workers finds a positive effect of the minimum wage on the gender gap (Cerejeira et al., 2012).





of the gender wage gap on top of the unavoidable pay raise by substituting the discrimination premium for the minimum-wage-pay-hike, shrinking that discrimination component.

4.3.2. Education

Another important aspect to be studied is the impact of the minimum wage for workers with different levels of education. As we discussed in the introduction, education was probably the most important driver of wage inequality dynamics over the last decades in Portugal. Increasing demand for skilled-workers in the 1980s and 1990s sharply inflated the returns to education, especially for higher-wage workers. When people began to get more educated, some workers started to detach from the generally uneducated majority of the workforce, increasing wage inequality at the top (Machado and Mata, 2005). But as more and more people obtained better education, that growth in wage inequality was mitigated (Centeno and Novo, 2014; Pereira, 2020).

However, since the mid-2000s, real wage growth was almost inversely proportional to education level: as we can see in figure 8, the average real wage growth of workers with the 1st and 2nd Cycles of education was 13% and 17%, respectively, while workers with secondary and higher education saw an average decrease in real wages of 3% and 14%.



 $\blacksquare Underlying change \qquad \equiv Change due to MW$

Figure 8: Average wage growth by education level, 2006-2019. This figure depicts the average real wage growth by level of education, and the part of that growth that can be attributed to the rise in the minimum wage. In the Portuguese education system the 1st and 2nd Cycles correspond to basic education, the 3rd Cycle to lower-secondary education, Secondary to upper-secondary education, and Higher to higher education, according to the ISCED classification.

The rise in the minimum wage was, in part, responsible for that inequality decreasing phenomenon, since it managed to increase more the wages of lower-educated workers relative to higher-educated ones. For instance, almost all of the 13% wage growth for





workers with the 1st Cycle was due to the rise in the minimum wage, while workers with higher education saw no direct effect in their paycheck from the rise in the minimum, on average.

Understanding the effect of the minimum wage on the returns to education would also be of interest - unlike for the gender wage gap, the literature on the effect of the minimum wage on the returns to education is very limited (Funkhouser, 1998 Bárány, 2016). While the wage equation coefficient estimates presented in table 4 are subject to a degree of endogeneity that does not allow us to infer causality, it may at least point us in some direction. In general, we can see that the wage premia coefficients from getting higher levels of education than the 1st cycle - the base level - all decreased between 2006 and 2019, and in all cases the decrease would have been lower, had the minimum wage not risen.

In sum, while education served to increase wage inequality in the past, between 2006 and 2019 lower-educated workers saw higher real wage gains than higher educated ones, and the minimum wage played a very important role in that phenomenon. The rise in the minimum wage actually seems to have lowered the returns to education, even if it was not the most important factor in their decrease.

But more than just decreasing inequality between workers with different levels of education, the minimum wage was also able to reduce wage inequality within education levels. As we can see in figure 9, wage inequality within education levels was much higher in the 2019 counterfactual scenario where the minimum wage stayed at its 2006 level than the "real" 2019, especially for the lower education levels - the rise in the minimum wage reduced the standard deviation of log wages among workers with the 1st cycle of education by 7 points, but only reduced it by 2 points among higher educated workers.



Actual 2019 Counterfactual 2019 with 2006 MW

Figure 9: Wage inequality by education level, 2019. This figure depicts the standard deviation of log wagesby level of education, as well as the counterfactual where the minimum wage stayed at its 2006 level.





4.3.3. Age

Regarding individual characteristics, it would also be useful to investigate how the minimum wage affected workers of different ages, the most renowned proxy for labour marketexperience. Figure 10 plots the average wage growth between 2006 and 2019 by years of age, as well as the part of that growth that can be attributed to the rise in the minimum wage.



Figure 10: Average wage growth by age, 2006-2019. This figure depicts the moving average of real wage growth by years of age, and the part of that growth that can be attributed to the rise in the minimum wage.

From 2006 to 2019, the wages that grew the most across the age spectrum were the wages of young workers in their 20s, and the wages of adults in their late-30s and 40s. However, the share of that growth that can be attributed to the rising minimum wage was very disparate - for a 20 year old worker, 18 of the 24 log point wage growth was due to the minimum wage (75%), while for a 40 year old worker, that was only 3 of the 18 points (17%).

It is expected that younger workers would be the ones benefiting the most from minimum wage hikes, since they are traditionally over-represented in the low-wage population due to their lack of work experience - and that is exactly what happened. Still, the impact of the minimum wage was not straightly decreasing across the age spectrum - it is decreasing at first, but starts to increase for older workers - for 55-year-olds and older, almost all wage growth was due to the rise in the minimum wage. That happens because, in Portugal, older workers tend to have lower wages than middle-aged workers, on average, due to the very strong dynamics in levels of education over time. These composition effects result in a U-shaped minimum wage effect curve across the age range.

4.3.4. Firms



Lastly, we may also look at the effects of the minimum wage on workers in different typesof firms. Figure 11 portrays the average wage growth, and the part of that growth the can be attributed to the rise in the minimum wage, by firm size.



■ Underlying change ■ Change due to MW

Figure 11: Average wage growth by firm size, 2006-2019. This figure depicts the average real wage growth by firm size, in number of employees, and the part of that growth that can be attributed to the rise in the minimum wage. The firm sizes in number of employees are: micro (1-9), small (10-49), medium (50-249) and large firms (+250).

In general, wages tend to grow with firm size, and the incidence of the minimum wage tends to decrease. That would lead us to believe that the impact of the minimum wage on average wages would be negatively correlated with firm size. However, it was significantly greater for workers in micro firms and large firms, while the average impact on workers insmall and medium firms was negligible - the rise in the minimum wage was responsible for43% of the 23 point rise in the average wage of workers in micro firms and was able to fully mitigate the 15 point decline in the average wage of workers in large firms.

This is because wage inequality is also greater within workers that work in larger firms, as figure 12 reports. The wage of workers in micro firms is low, on average, and concentrated at the bottom, while the average wage of workers in large firms is higher, but it is so because of a larger number of high-earning workers, since large firms also concentrate a very significant share of low-wage workers. That is why the rise in the minimum wage was responsible for a large increase in average wages of workers in these firms, as well as a larger reduction in inequality between these workers. On the contrary, small and medium firms have maintained relatively higher wages for the majority of their workers, not being as sensitive to changes in the minimum wage. Another reason for this phenomenon could be the larger collective agreement coverage and union power among large-firm-employees (Addison et al., 2022), which tend to be renegotiation at a broad scale whenever there is a minimum wage hike.

Another way to divide firms is by their productivity. In figure 13 we proxy the labour productivity of a firm as its sales divided by the number of employees. As we can see,





the average growth of wages between 2006 and 2019 was stable across the productivity distribution, around 20%, except for workers at the 20% less productive and 20% more



Actual 2019 Counterfactual 2019 with 2006 MW

Figure 12: Wage inequality by firm size, 2019. This figure depicts the standard deviation of log wages by firm size, and the counterfactual where the minimum wage stayed at its 2006 level.





Figure 13: Average wage growth by firm's labour productivity percentile, 2006-2019. This figure depicts the average real wage growth of workers by their firms' percentile in the labour productivity distribution, and the wage growth that can be attributed to the change in the minimum wage. A firm's labour productivity is measured as sales divided by the number of employees.

At the very least productive firms, wages actually went down. It is, of course, expected that less productive firms have less leverage to raise their workers' wages, but the minimumwage was still able to prevent much larger pay-cuts at these firms. That may have even triggered the phenomenon that we discussed in section 2, confirmed by



Alexandre et al. (2021), whereby a number of unproductive firms are likely going under - as they cannot support the extra labour costs - increasing the average productivity of the economy. For relatively more productive firms, at the second and third quintiles, the minimum wage wasstill able to generate wage gains at a visible rate, but for workers at more productive firms, the rise in minimum wage was not very significant - not being able to prevent the much slower wage growth at the most productive firms.

4.4. Total wages and base wages

All prior analysis was done by looking at the total monthly wage - composed by thebase wage and other regular and irregular benefits¹² - which is the most relevant in the measurement of inequality. But the distributions of base and total wages have presented rather different shapes, as portrayed by the histograms in figure 3. And figure 2 had already hinted that base and total wages were not reacting to changes in the minimum wage in thesame manner since 2006, as the share of workers on the minimum wage has been rapidly increasing if we look at base wages, but if we look at total wages it actually decreased.

Figure 14 depicts the evolution of the average difference between total and base wages the total-base wage differential - at different points of the distribution. As can see, the total wages of middle- and low-wage workers have been increasing at a pace that is not reflected in their base wages. This phenomenon has increased the total-base wage differential across the distribution over the years, except at the top, where it remained stable.



Figure 14: Evolution of the total-base wage differential at different points of the distribution. This figuredepicts the average log differential between total and base wages at the 10th, 50th and 90th percentiles.

¹² Regular benefits include food/ rent/ transport allowances, seniority/ tenure/ productivity/ assiduity bonus and allowances for heavy, dangerous or nightly work. Irregular benefits include payment of profits, securities, indemnities, retroactive, assiduity and productivity bonus, and Christmas and Holidays bonus.





However, since the 2000s, that differential ceased to increase at the very bottom of the distribution. One driver of that phenomenon could be the swiftly rising minimum wage, which started catching up and pushing up the slower-growing base wages of more and more workers at the bottom of the distribution, where the minimum wage bound, as figure 15 illustrates. Since total wages kept their pace of growth, the differential stopped growing.



Figure 15: Evolution of base and total wages for different percentiles at the bottom of the distribution. This figure depicts the average wage at different percentiles at the lower-end of the distribution for base and total wages, as well as the minimum wage line.

In order to understand how the minimum wage participated in this development, the actual and counterfactual wage densities were estimated, but using base wages instead of total wages - figure 16. The counterfactual wage density in panel A is not of straightforwardinterpretation, but if we compare the change in density that can be attributed to rise inthe minimum wage for total and base wages, in panel B, we can see that the effect of the minimum wage on the distribution of base wages was much more concentrated at the newminimum, while total wages experienced considerably larger spillover effects.

This suggests that, while minimum wage bite effects are widely reflected in the base wage, spillover effects predominantly manifest themselves in the other benefits that make up the total wage. Since 2006, a very significant share of workers at the bottom of the distribution saw their wages increase due to the rise in the minimum wage. By looking at base wages it would appear that the minimum was only pushing up the wages of the workers for whom the minimum bound, but when we look at the effect on total wages, we can understand that spillovers greatly magnified the effect of the minimum wage, pushing an extremely significant mass of wages higher up in the pay-scale. While spillover effects were minimal in base wages, they were much greater when we look at the entirety wages.

This was quantified in table 5. Just like for total wages, we see a great impact of the minimum wage in reducing base wage inequality, although in this case it is almost all due to the minimum wage bite - spillovers are minor, and only reach up to 10% above the



Gabinete de Estratégia e Estudos

minimum. For instance, the bite single handedly cut the 90-10 percentile differential by 21 points, while spillovers cut it by only 1 point. The bite also raised the wages of the 8% of workers forwhom the minimum wage bound in 2006, and was responsible for one-third of the 12 point increase in the mean wage.



Figure 16: Density estimates using base wages. This figure depicts the density estimate graph previously presented, but using base wages instead of total wages. Panel A depicts the densities analogously to figure4, and Panel B depicts the change in density reported in Panel C of figure 5 and the analogous change in density using base wages.

Still, the influence of the minimum on the distribution is visibly smaller for base wages, especially due to the lack of spillovers: while the share of workers earning more than the real value of the 2019 minimum wage grew between 2006 and 2019, for both base (58% to 79%) and total wages (77% to 96%), the share of that increase thatcan be attributed to the minimum is only one-third in the case of base wages, much lessthan the two-thirds we found for total wages.

This difference may look like an accounting detail but it is rather important since both public debate and academic research often focus solely on base wages when addressing the impacts of the minimum wage - for instance, the incidence of the minimum wage has been rapidly approaching one fourth of the the workforce, looking at base wages, a figure which frequently makes newspaper covers. However, that figure hides a more intricate reality whereby a significant share of minimum wage effects are materialising in total wages and not on base wages.

Still, the reasons for why employers have been compensating their employees with other benefits is not clear. Although it is reasonable that, when the minimum wage increases and employers feel the need to compensate relatively more productive or more senior employees, they do it through benefits like productivity or seniority bonuses, other plausible reasons come to mind. It could be happening as employers shield themselves from extreme nominal wage rigidity in the base wage (Carneiro et al., 2014; Guimarães et al., 2017) that may not reflect in the other components of wages, because





	84	00.10	00.50	50.10	Incidence	Incidence	Mean
	bu	90-10	90-00	50-10	1st year	last year	wage
		Base	e wages				
2006	0.51	1.24	0.87	0.37	0.08	0.42	5.35
2019	0.46	1	0.83	0.17	0	0.21	5.47
2019 w/ 2006 MW	0.5	1.22	0.84	0.38	0.08	0.29	5.42
2019 w/o spillovers	0.46	1.01	0.86	0.15	0	0.25	5.46
Total change	-0.05	-0.24	-0.04	-0.2	-0.08	-0.21	0.12
Underlying change	-0.01	-0.02	-0.03	0.01	0	-0.13	0.07
Change due to MW	-0.04	-0.22	-0.01	-0.21	-0.08	-0.08	0.05
Change due to bite	-0.04	-0.21	0.02	-0.23	-0.08	-0.04	0.04
Change due to spillovers	0	-0.01	-0.03	0.02	0	-0.04	0.01

of the decline in unionisation, since unions tended to focus collective agreements negotiations on base wages, which employers partly offset in other benefits

Table 5: Decomposition of changes in statistics of interest using base wages, 2006-2019. This table depicts the statistics of interest, analogously to table 2, but using base wages instead of total wages.

(Cardoso and Portugal, 2005; Addison et al., 2022), or simply due to tax optimisation by firms, since not all the benefits that compose the differential aresubject to the same tax rules. Unfortunately, the data collected in *Quadros de Pessoal* do not discriminate these components.

5. Conclusion

Over the last three-and-a-half decades, wage inequality in Portugal increased, then it stagnated, and then it fell. And the importance of the minimum wage in the labour market presented a nearly symmetric behaviour. This paper explored this significant negative correlation to try and understand the impact of the minimum wage on the whole shape of the wage distribution.

The results visually and quantitatively show how much the minimum wage can structurally reshape the wage distribution. Between 2006 and 2019, the remarkable rise in the real minimum wage was responsible for virtually all of the decrease in wage inequality in Portugal, especially by deterring an increase in inequality at the lower-tail of the distribution. And most of that impact was often driven by spillover effects, since workers' wages did not simply accumulate at the minimum - they cascaded up to 40% above the minimum wage. More, the minimum wage was not only able to structurally reshape the wage distribution, but it actually shifted it to the right, explaining 40% of the Portuguese average wage growth over that period.

In accordance with the literature, the minimum wage was most important in reducing wage inequality between women, which would have actually increased had the minimumwage not risen, and between genders, being responsible for 60% of the female average wage growth (just 7% for men), and for a reduction of 5 p.p., or a quarter of the gender wage gap.It



was also highly important in reducing within and between wage inequality in regards to education, explaining almost all of the wage growth of uneducated workers and none of the wage change for higher educated workers, and showing signs of a potential mitigating effecton the returns to education. The minimum wage also benefited younger workers the most, along with older workers approaching retirement age. By looking at firm characteristics, wesee that the minimum wage was most impactful on workers at micro and large firms, and on workers at less productive firms, displaying a possible productivity inducing effect.

The analysis was mostly done using total wages, but looking at base wages uncovered a rather interesting phenomenon. While minimum wage bite effects were reflected in the base wage, spillover effects predominantly manifested themselves in fringe benefits and other components of the total wage. Since spillovers embody an extremely significant share of the effects of the minimum wage on the wage distribution, focusing only on the base wages - which frequently happens both in academic research and in the public debate - hides a substantial part of the true impact of the minimum wage.

Such a powerful reshaping of the wage distribution, such high spillovers, or such an impact on the average wage are not common in the literature, but neither is a compression of the distribution and a minimum wage hike like these. Exploring that phenomenon, this article adds several insights to the existing minimum wage research, laying the groundwork for other issues to be addressed in future research. Via an alternative approach, this article notonly addresses the proposed research question with meaningful results - yes, the minimum wage does reshape the wage distribution - but it also discloses various paths through whichminimum wage externalities may arise - like the compression of the gender gap and education premium, the impact on labour costs of unproductive firms, or its magnifying effect on fringe benefits.

Appendix A. The reweighted adaptive kernel density estimator

The reweighted adaptive kernel density estimator applied in the construction of counterfactual wage densities is defined as

$$\hat{f}_{h}(y) = \frac{1}{hN\lambda_{i}}\sum_{i=1}^{n}\psi(X_{k}) \cdot K\left(\frac{y-Y_{i}}{h\lambda_{i}}\right)$$
(A.1)

where K(.) is an alternative Epanechnikov kernel function, h is the Silverman (1986) bandwidth, and λ_i are the local bandwidth factors (Abramson, 1982; Jann, 2007), based on a preliminary fixed bandwidth density estimate, and estimated as

$$\hat{\lambda}_{i} = \sqrt{\frac{G(\hat{f}_{h}(Y))}{\hat{f}_{h}(Y)}}, \quad i = 1, \cdots, n.$$
 (A.2)

where G(.) stands for the geometric mean over all *i*. Note that $G(\lambda) = 1$, and thus





 $G(h\lambda) = h.$

Then, $\psi(X_k)$ is the reweighting factor proposed in DiNardo et al. (1996) and generalized in Fortin et al. (2011), estimated as

$$\hat{\psi}(X_k) = \frac{\bar{Y}_k^C(X)}{\bar{Y}_k(X)}, \quad k = 1, \cdots, K.$$
(A.3)

Appendix B. Interpreting the decomposition tables and results

Tables 2, 3 and 5 present some of the main quantitative results of this paper. The firstfour rows of each panel present statistics of interest for the different scenarios that were estimated. Using panel A of table 2 as reference: *1986*, the actual statistics of the first year; *1994*, the actual statistics of the last year; *1994 w/ 1986 MW*, the counterfactual statistics of the last year; *the last year*, had the minimum wage stayed at its first year level; *1994 w/ spillovers*, the counterfactual statistics of the last year, had there been no minimum wage spillover effects.

The three rows after that present the decomposed change in those statistics. *Total change* is the actual change that occurred between 1986 and 1994, *underlying change* is the change that would have occurred between those years if the real value of the minimum wage had stayed at its 1986 level, and the *change due to the minimum wage* is the change driven by the shift in the minimum wage between those years:

Total change = Underlying change + Change due to MW Underlying change = 1994 w/ 1986 MW - 1986 (B.1) Change due to MW = 1994 - 1994 w/ 1986 MW

The two final rows of each panel present the *Change due to MW* decomposed into the *change due to the minimum wage bite*, the change driven by the workers whose wages would be below the new minimum but the minimum wage managed to push to that new minimum, and the *change due to the spillovers*, driven by the workers who would be earning the new minimum wage or more but, due to the minimum wage raise, still benefited from a wage boost:

Change due to MW = Change due to bite + Change due to spilloversChange due to bite = 1994 w/o spillovers - 1994 w/ 1986 MW (B.2) Change due to spillovers = 1994 - 1994 w/o spillovers



Appendix C. Tables

Year	1986	1994	2006	2019
Mean wage	5.24	5.42	5.57	5.73
Mean wage (base wage)	5.10	5.24	5.35	5.47
Standard deviation	0.50	0.57	0.58	0.51
Standard deviation (base wage)	0.40	0.49	0.51	0.46
90-10 percentile differential	1.21	1.41	1.43	1.21
90-50 percentile differential	0.79	0.92	0.95	0.86
75-25 percentile differential	0.66	0.77	0.76	0.65
50-10 percentile differential	0.42	0.49	0.48	0.35
Minimum wage	4.72	4.73	4.82	5.11
Minimum wage relative to the mean	0.66	0.59	0.54	0.59
Incidence of the minimum wage	0.11	0.06	0.04	0.04
Incidence of the minimum wage (base wage)	0.14	0.09	0.08	0.21

Table C.6: Descriptive distributional statistics. This table provides descriptive measures of log total real monthly wages in Portugal (unless specified) for the four years that divide the three periods of analysis, namely on mean wages wage inequality, and the minimum wage.

Year	Id	Log real wage	Year	Id	Log real wage	Wage bin	Outcome
2010	3825	7.26	2010	3825	7.26	55	1
2010	3836	4.73	2010	3825	7.26	56	1
			2010	3825	7.26	57	0
			2010	3825	7.26	58	0
			2010	3825	7.26	59	0
			2010	3825	7.26	60	0
			2010	3825	7.26	61	0
			2010	3825	7.26	62	0
			2010	3836	4.73	1	1
			2010	3836	4.73	2	1
			2010	3836	4.73	3	1
			2010	3836	4.73	4	1
			2010	3836	4.73	5	1
			2010	3836	4.73	6	0
			2010	3836	4.73	7	0

Table C.7: Illustration of the stacked dataset. This table illustrates how the original dataset was stacked in order to perform the estimation. Each observation in the original data was multiplied by 62, the number of wage bins, and the outcome variable was determined for each wage bin, so as to "simultaneously estimate62 probit regressions".

Appendix D. Figures







Figure D.17: Counterfactual densities in the absence of spillovers for 1986-1994, 1994-2006, and 2006-2019. Each panel of this figure depicts the counterfactual density estimate for the last year of each period of analysis, had there been no minimum wage spillover effects, as well as the actual density estimate for that year. The dashed vertical line indicates minimum wage.





A) 1986-1994







Figure D.18: Wage growth incidence curve over the three periods of analysis. This graph depicts the actual and counterfactual growth rate of wages by percentile of the distribution between 1986-1994, 1994-2006, and 2006-2019. The slight difference between the actual and counterfactual curves at higher percentiles of the distribution is due to the reweighing, not representing actual lower wage growth.





References

- Abramson, I.S., 1982. On bandwidth variation in kernel estimates-a square root law. The Annals of Statistics 10, 1217 1223. doi:10.1214/aos/1176345986.
- Addison, J.T., Portugal, P., de Almeida Vilares, H., 2022. Union membership density and wages: The role of worker, firm, and job-title heterogeneity. Journal of Econometrics doi:10.1016/j.jeconom.2021.12.002.
- Addison, J.T., Portugal, P., Vilares, H., 2017. Unions and collective bargaining in the wake of the greatrecession: evidence from portugal. British Journal of Industrial Relations 55, 551–576. doi:10.1111/bjir.12198.
- Aeberhardt, R., Givord, P., Marbot, C., 2016. Spillover effect of the Minimum Wage in France: An Unconditional Quantile Regression. Working Paper 2016-05. Center for Research in Economics and Statistics. URL: https://ideas.repec.org/p/crs/wpaper/2016-05.html.
- Alexandre, F., Ba ção, P., Cerejeira, J., Costa, H., Portela, M., 2021. Minimum wage and financially distressed firms: another one bites the dust. Labour Economics, 102088URL: <u>https://</u> www.sciencedirect.com/science/article/pii/S0927537121001238, doi:10.1016/j.labeco.2021. 102088.
- Alvaredo, F., 2009. Top incomes and earnings in portugal 1936–2005. Explorations in Economic History 46, 404–417. doi:10.1016/J.EEH.2009.04.006.
- Angrist, J.D., Pischke, J.S., 2010. The credibility revolution in empirical economics: How better research design is taking the con out of econometrics. Journal of Economic Perspectives 24, 3–30. doi:10.1257/jep.24.2.3.
- Autor, D.H., Katz, L.F., Kearney, M.S., 2008. Trends in u.s. wage inequality: Revising the revisionists. The Review of Economics and Statistics 90, 300–323. URL: http://www.mitpressjournals.org/, doi:10. 1162/REST.90.2.300.
- Autor, D.H., Manning, A., Smith, C.L., 2016. The contribution of the minimum wage to us wage inequality over three decades: A reassessment. American Economic Journal: Applied Economics 8, 58–99. doi:10. 1257/APP.20140073.
- Bárámy, Z.L., 2016. The minimum wage and inequality: the effects of education and technology. Journal of Labor Economics 34, 237–274. doi:10.1086/682346.
- Bargain, O., Doorley, K., Van Kerm, P., 2019. Minimum wages and the gender gap in pay: new evidence from the united kingdom and ireland. Review of Income and Wealth 65, 514–539. doi:10.1111/roiw.12384.
- Belman, D., Wolfson, P.J., 2014. What does the minimum wage do? WE Upjohn Institute. doi:10.1057/ eej.2015.25.





- Bossler, M., Schank, T., 2020. Wage Inequality in Germany after the Minimum Wage Introduction. Technical Report 13003. IZA Institute of Labor Economics. URL: https://www.iza.org/publications/dp/ 13003/wage-inequality-in-germany-afterthe-minimum-wage-introduction.
- Bound, J., Johnson, G., 1992. Changes in the structure of wages in the 1980's: an evaluation of alternative explanations. American Economic Review 82, 371–392. doi:10.2307/2117311.
- Brown, C., 1999. Minimum wages, employment, and the distribution of income, in:
 Ashenfelter, O., Card, D. (Eds.), Handbook of Labor Economics. Elsevier. volume
 3B. chapter 32, pp. 2101–2163. doi:10.1016/S1573-4463(99)30018-3.
- Brown, C., Gilroy, C., Kohen, A., 1982. The effect of the minimum wage on employment and unemployment: a survey. Journal of Economic Literature 20, 487–528. doi:10.3386/w0846.
- Butcher, T., Dickens, R., Manning, A., 2012. Minimum Wages and Wage Inequality: Some Theory and an Application to the UK. CEP-LSE Working Paper Series 4512. Centre for Economic Performance, LSE.
- Caliendo, M., Wittbrodt, L., 2021. Did the Minimum Wage Reduce the Gender Wage Gap in Germany?Technical Report 14926. IZA Institute of Labor Economics.
- Campos Lima, M.d.P., Martins, D., Costa, A.C., Velez, A., 2021. Internal devaluation and economic inequality in portugal: challenges to industrial relations in times of crisis and recovery. Transfer: European Review of Labour and Research 27, 47–73. doi:10.1177/1024258921995006.
- Card, D., Cardoso, A.R., Kline, P., 2016. Bargaining, sorting, and the gender wage gap: Quantifying the impact of firms on the relative pay of women. The Quarterly Journal of Economics 131, 633–686.doi:10.1093/qje/qjv038.
- Card, D., DiNardo, J.E., 2002. Skill-biased technological change and rising wage inequality: Some problems and puzzles. Journal of Labor Economics 20, 733–783. doi:10.1086/342055.
- Card, D., Krueger, A.B., 1994. Minimum wages and employment: a case study of the fast-food industry in new jersey and pennsylvania. American Economic Review 84, 772–793. doi:10.1257/aer.90.5.1397.
- Card, D., Krueger, A.B., 1995. Time-series minimum-wage studies: A meta-analysis. American EconomicReview 85, 238–243.
- Card, D., Krueger, A.B., 2015. Myth and Measurement: The New Economics of the Minimum Wage. Princeton University Press. doi:10.2307/2234949.
- Cardoso, A.R., 1997. Workers or employers: Who is shaping wage inequality? Oxford Bulletin of Economics and Statistics 59, 523–547. doi:10.1111/1468-0084.00081.





- Cardoso, A.R., 1998. Earnings inequality in portugal: High and rising? Review of Income and Wealth 44,325–343. doi:10.1111/J.1475-4991.1998.TB00285.X.
- Cardoso, A.R., 1999. Firms' wage policies and the rise in labor market inequality: The case of Portugal. Industrial and Labor Relations Review 53, 87–102. doi:10.2307/2696163.
- Cardoso, A.R., Portugal, P., 2005. Contractual wages and the wage cushion under different bargaining settings. Journal of Labor Economics 23, 875–902. doi:10.1086/491608.
- Carneiro, A., Portugal, P., Varejão, J., 2014. Catastrophic job destruction during the portuguese economic crisis. Journal of Macroeconomics 39, 444–457. doi:10.1016/j.jmacro.2013.09.018.
- Carneiro, A., S´a, C.A.d.S.P., Cerejeira, J., Varejão, J., Portela, M., et al., 2011. Estudo sobre a retribuição m´ınima mensal garantida em Portugal. Technical Report. Minist´erio da Economia e do Emprego. URL: http://hdl.handle.net/1822/14227.
- Cengiz, D., Dube, A., Lindner, A., Zipperer, B., 2019. The effect of minimum wages on low-wage jobs. The Quarterly Journal of Economics 134, 1405–1454. doi:10.1093/qje/qjz014.
- Centeno, M., Novo, A.A., 2014. When supply meets demand: wage inequality in portugal. IZA Journal of European Labor Studies 3, 1–20. doi:10.1186/2193-9012-3-23.
- Cerejeira, J., Kizilca, F.K., Portela, M., S´a, C., 2012. Minimum wage, fringe benefits, overtime payments and the gender wage gap. Technical Report. IZA Institute of Labor Economics.
- Chernozhukov, V., Fern andez-Val, I., Melly, B., 2013. Inference on counterfactual distributions. Econometrica 81, 2205–2268. doi:10.3982/ECTA10582.
- Cho, J., Yang, D., 2021. Impact of minimum wage legislation on gender differentials in sticky wages: evidence from korea. Applied Economics Letters doi:10.1080/13504851.2021.2000929.
- DiNardo, J., Fortin, N.M., Lemieux, T., 1996. Labor market institutions and the distribution of wages, 1973-1992: A semiparametric approach. Econometrica 64, 1001–1044. URL: http://www.jstor.org/ stable/2171954, doi:2171954.
- Dolado, J., Kramarz, F., Machin, S., Manning, A., Margolis, D., Teulings, C., 1996. The economic impactof minimum wages in europe. Economic Policy 11, 317–372. doi:1344707.





- Dube, A., Lester, T.W., Reich, M., 2010. Minimum wage effects across state borders: Estimates using contiguous counties. Review of Economics and Statistics 92, 945– 964. doi:10.1162/REST_a_00039.
- Dustmann, C., Lindner, A., Schönberg, U., Umkehrer, M., vom Berge, P., 2022. Reallocation effects of the minimum wage*. The Quarterly Journal of Economics 137, 267–328. doi:10.1093/gje/gjab028.
- Feenstra, R.C., Hanson, G.H., 1996. Globalization, outsourcing, and wage inequality. American Economic Review 86, 240–245. doi:10.3386/w5424.
- Fortin, N.M., Lemieux, T., 1998. Rank regressions, wage distributions, and the gender gap. Journal of Human Resources 33, 610–643. doi:10.2307/146335.
- Fortin, N.M., Lemieux, T., Firpo, S., 2011. Decomposition methods in economics, in: Ashenfelter, O., Card, D. (Eds.), Handbook of Labor Economics. Elsevier. volume 4A. chapter 1, pp. 1–102. doi:10. 1016/S0169-7218(11)00407-2.
- Fortin, N.M., Lemieux, T., Lloyd, N., 2021. Labor market institutions and the distribution of wages: The role of spillover effects. Journal of Labor Economics 39, S369–S412. doi:10.1086/712923.
- Freeman, R.B., 1996. The minimum wage as a redistributive tool. The Economic Journal 106, 639–649. doi:10.2307/2235571.
- Frölich, M., 2004. Finite-sample properties of propensity-score matching and weighting estimators. The Review of Economics and Statistics 86, 77–90. doi:10.1162/003465304323023697.
- Funkhouser, E., 1998. Changes in the returns to education in costa rica. Journal of Development Economics 57, 289–317.
- Gramlich, E.M., 1976. Impact of minimum wages on other wages, employment, and family incomes. Brookings Papers on Economic Activity 1976, 409–462. doi:10.2307/2534380.
- Grossman, J.B., 1983. The impact of the minimum wage on other wages. The Journal of Human Resources 18, 359–378. doi:10.2307/145206.
- Guimarães, P., Martins, F., Portugal, P., 2017. Upward Nominal Wage Rigidity.
 Working Papers 2017.Banco de Portugal Economics and Research Department.
 doi:10.2139/ssrn.2911408.
- Hoffmann, F., Lee, D.S., Lemieux, T., 2020. Growing income inequality in the united states and other advanced economies. Journal of Economic Perspectives 34, 52–78. URL: https://www.aeaweb.org/articles?id=10.1257/jep.34.4.52, doi:10.1257/jep.34.4.52.





- Jann, B., 2007. KDENS: Stata module for univariate kernel density estimation. Statistical Software Components, Boston College Department of Economics. URL: https://ideas.repec.org/c/boc/bocode/ s456410.html.
- Joe, D.H., Moon, S., 2020. Minimum wages and wage inequality in the oecd countries. East Asian Economic Review 24, 253–273. doi:10.11644/kiep.eaer.2020.24.3.379.
- Katz, L.F., Krueger, A.B., 1992. The effect of the minimum wage on the fast-food industry. Industrial and Labor Relations Review 46, 6–21. doi:10.1177/001979399204600102.
- Koeniger, W., Leonardi, M., Nunziata, L., 2007. Labor market institutions and wage inequality. Industrial and Labor Relations Review 60, 340–356. doi:10.1177/001979390706000302.
- Lee, D.S., 1999. Wage Inequality in the United States During the 1980s: Rising Dispersion or Falling Minimum Wage? The Quarterly Journal of Economics 114, 977–1023. URL: https://academic.oup. com/qje/article-pdf/114/3/977/5228618/114-3-977.pdf, doi:10.1162/003355399556197.
- Li, S., Ma, X., 2015. Impact of minimum wage on gender wage gaps in urban china. IZA Journal of Labor & Development 4. doi:10.1186/s40175-015-0044-4.
- Machado, J.A.F., Mata, J., 2005. Counterfactual decomposition of changes in wage distributions using quantile regression. Journal of Applied Econometrics 20, 445–465. URL: https://onlinelibrary. wiley.com/doi/10.1002/jae.788, doi:10.1002/JAE.788.
- Machin, S., 1997. The decline of labour market institutions and the rise in wage inequality in Britain. European Economic Review 41, 647–657. doi:10.1016/S0014-2921(97)00027-5.
- Machin, S., Manning, A., 1994. The effects of minimum wages on wage dispersion and employment: Evidence from the u.k. wages councils. ILR Review 47, 319–329. doi:10.1177/001979399404700210.
- Machin, S., Manning, A., Rahman, L., 2003. Where the minimum wage bites hard: Introduction of minimum wages to a low wage sector. Journal of the European Economic Association 1, 154–180. doi:10.1162/154247603322256792.
- Majchrowska, A., Strawiń ski, P., 2018. Impact of minimum wage increase on gender wage gap: Case of poland. Economic Modelling 70, 174–185. doi:10.1016/j.econmod.2017.10.021.
- Manning, A., 2003. Monopsony in motion: Imperfect competition in labor markets. Princeton University Press. doi:10.2307/4126626.





- Martins, P.S., 2021. 30,000 minimum wages: The economic effects of collective bargaining extensions. British Journal of Industrial Relations 59, 335–369. doi:10.1111/bjir.12564.
- Martins, P.S., Pereira, P.T., 2004. Does education reduce wage inequality? quantile regression evidence from 16 countries. Labour Economics 11, 355–371. doi:10.1016/j.labeco.2003.05.003.
- Meyer, R.H., Wise, D.A., 1983a. Discontinuous distributions and missing persons: The minimum wage and unemployed youth. Econometrica 11, 1677–1698. doi:doi.org/1912112.
- Meyer, R.H., Wise, D.A., 1983b. The effects of the minimum wage on the employment and earnings of youth. Journal of Labor Economics 1, 66–100. doi:10.1086/298005.
- Milanovic, B., 2016. Global Inequality: A New Approach for the Age of Globalization. Harvard University Press, Cambridge, MA. doi:10.4159/9780674969797.
- Mincer, J., 1976. Unemployment effects of minimum wages. Journal of Political Economy 84, S87–S104. doi:10.1086/260534.
- Neumark, D., Schweitzer, M., Wascher, W., 2004. Minimum wage effects throughout the wage distribution. Journal of Human Resources 39, 425–450. doi:10.2307/3559021.
- Oliveira, C., 2021. How is the Minimum Wage Shaping the Wage Distribution: Minimum Wage, Spillovers and Wage Inequality in Portugal. Master's thesis. Nova School of Business and Economics. Lisbon, Portugal. doi:10.13140/RG.2.2.23479.19362.
- Pereira, J.M., 2020. Did wage inequality increase in portugal? yes, and for good reasons. Applied Economics Letters 28, 973–977. doi:10.1080/13504851.2020.1789057.
- Pereira, J.M., Galego, A., 2019. Diverging trends of wage inequality in europe. Oxford Economic Papers 71, 799–823. URL: https://academic.oup.com/oep/article/71/4/799/5307420, doi:10.1093/OEP/ GPY072.
- Pereira, S.C., 2003. The impact of minimum wages on youth employment in portugal. European EconomicReview 47, 229–244. doi:10.1016/S0014-2921(02)00209-X.
- Portugal, P., Cardoso, A.R., 2006. Disentangling the minimum wage puzzle: An analysis of worker accessions and separations. Journal of the European Economic Association 4, 988–1013. doi:10.1162/JEEA.2006. 4.5.988.
- Portugal, P., Raposo, P.S., Reis, H., 2018. The distribution of wages and wage inequality. Technical Report. Banco de Portugal Economics and Research Department.





- Rebitzer, J.B., Taylor, L.J., 1995. The consequences of minimum wage laws some new theoretical ideas. Journal of Public Economics 56, 245–255. doi:10.1016/0047-2727(93)01411-3.
- Robinson, H., 2002. Wrong side of the track? the impact of the minimum wage on gender pay gaps inbritain. Oxford Bulletin of Economics and Statistics 64, 417–448. doi:10.1111/1468-0084.00271.
- Rubery, J., Grimshaw, D., 2011. Gender and the minimum wage, in: Regulating for decent Work. Springer, pp. 226–254. doi:10.1057/9780230307834_9.
- Shannon, M., 1996. Minimum wages and the gender wage gap. Applied Economics 28, 1567–1576. doi:10.1080/000368496327543.
- Silverman, B.W., 1986. Density estimation for statistics and data analysis. Chapman & Hall, New York.doi:10.1201/9781315140919.
- Stewart, M.B., 2012. Wage inequality, minimum wage effects, and spillovers. Oxford Economic Papers 64,616–634. doi:10.1093/oep/gps003.
- Stigler, G.J., 1946. The economics of minimum wage legislation. American Economic Review 36, 358–365. Teulings, C.N., 2003. The contribution of minimum wages to increasing wage inequality. The Economic Journal 113, 801–833. doi:10.1111/1468-0297.t01-1-00163.





GEE Papers

- 1: Evolução do Comércio Externo Português de Exportação (1995-2004) João Ferreira do Amaral
- 2: Nowcasting an Economic Aggregate with Disaggregate Dynamic Factors: An Application to Portuguese GDP Antonio Morgado | Luis Nunes | Susana Salvado
- 3: Are the Dynamics of Knowledge-Based Industries Any Different? Ricardo Mamede | Daniel Mota | Manuel Godinho
- 4: Competitiveness and convergence in Portugal Jorge Braga de Macedo
- 5: Produtividade, Competitividade e Quotas de Exportação Jorge Santos
- 6: Export Diversification and Technological Improvement: Recent Trends in the Portuguese Economy Manuel Cabral
- 7: Election Results and Opportunistic Policies: An Integrated Approach Toke Aidt | Francisco Veiga | Linda Veiga
- 8: Behavioural Determinants of Foreign Direct Investment Ricardo Pinheiro-Alves
- 9: Structural Transformation and the role of Foreign Direct Investment in Portugal: a descriptive analysis for the period 1990-2005 Miguel de Freitas | Ricardo Mamede
- 10: Productive experience and specialization opportunities for Portugal: an empirical assessment Miguel de Freitas | Susana Salvado | Luis Nunes | Rui Costa Neves
- 11: The Portuguese Active Labour Market Policy during the period 1998-2003 - A Comprehensive Conditional Difference-In-Differences Application Alcina Nunes | Paulino Teixeira
- 12: Fiscal Policy in a Monetary Union: Gains from Changing Institutions Susana Salvado
- 13: Coordination and Stabilization Gains of Fiscal Policy in a Monetary Union Susana Salvado
- 14: The Relevance of Productive Experience in the Process of Economic Growth: an Empirical Study Diana Vieira
- 15: Employment and Exchange rates: the Role of Openness and Technology Fernando Alexandre | Pedro Bação | João Cerejeira | Miguel Portela

- 16: Aggregate and sector-specific exchange rate indexes for the Portuguese economy Fernando Alexandre | Pedro Bação | João Cerejeira | Miguel Portela
- 17: The Macroeconomic Determinants of Cross Border Mergers and Acquisitions and Greenfield Investments Paula Neto | Antonio Brandao | António Cerqueira
- 18: Does the location of manufacturing determine service sectors' location choices? Evidence from Portugal Nuno Crespo | Maria Paula Fontoura
- 19: A hipótese do Investment Development Path: Uma Abordagem por Dados em Painel. Os casos de Portugal e Espanha Miguel Fonseca | António Mendonça | José Passos
- 20: Outward FDI Effects on the Portuguese Trade Balance, 1996-2007 Miguel Fonseca | António Mendonça | José Passos
- 21: Sectoral and regional impacts of the European Carbon Market in Portugal Margarita Robaina Alves | Miguel Rodriguez | Catarina Roseta-Palma
- 22: Business Demography Dynamics in Portugal: A Non-Parametric Survival Analysis Alcina Nunes | Elsa Sarmento
- 23: Business Demography Dynamics in Portugal: A Semi-parametric Survival Analysis Alcina Nunes | Elsa Sarmento
- 24: Digging Out the PPP Hypothesis: an Integrated Empirical Coverage Miguel de Carvalho | Paulo Júlio
- 25: Regulação de Mercados por Licenciamento Patrícia Cerqueira | Ricardo Pinheiro Alves
- 26: Which Portuguese Manufacturing Firms Learn by Exporting? Armando Silva | Óscar Afonso | Ana Paula Africano
- 27: Building Bridges: Heterogeneous Jurisdictions, Endogenous Spillovers, and the Benefits of Decentralization Paulo Júlio | Susana Peralta
- 28: Análise comparativa de sobrevivência empresarial: o caso da região Norte de Portugal Elsa Sarmento | Alcina Nunes
- 29: Business creation in Portugal: Comparison between the World Bank data and Quadros de Pessoal Elsa Sarmento | Alcina Nunes
- 30: The Ease of Doing Business Index as a tool for Investment location decisions



João Zambujal Oliveira | Ricardo Pinheiro Alves

- 31: The Politics of Growth: Can Lobbying Raise Growth and Welfare? Paulo Júlio
- 32: The choice of transport technology in the presence of exports and FDI José Pedro Ponte | Armando Garcia Pires
- 33: Tax Competition in an Expanding European Union

Ronald Davies | Johannes Voget

- 34: The usefulness of State trade missions for the internationalization of firms: an econometric analysis Ana Paula Africano | Aurora Teixeira | André Caiado
- 35: The role of subsidies for exports: Evidence from Portuguese manufacturing firms Armando Silva
- 36: Criação de empresas em Portugal e Espanha: análise comparativa com base nos dados do Banco Mundial Elsa Sarmento | Alcina Nunes
- 37: Economic performance and international trade engagement: the case of Portuguese manufacturing firms Armando Silva | Oscar Afonso | Ana Paula Africano
- 38: The importance of Intermediaries organizations in international R&D cooperation: an empirical multivariate study across Europe Aurora Teixeira | Margarida Catarino
- 39: Financial constraints, exports and monetary integration - Financial constraints and exports: An analysis of Portuguese firms during the European monetary integration Filipe Silva | Carlos Carreira
- 40: FDI and institutional reform in Portugal Paulo Júlio | Ricardo Pinheiro-Alves | José Tavares
- 41: Evaluating the forecast quality of GDP components
 Paulo Júlio | Pedro Esperança | João C. Fonseca
- 42: Assessing the Endogeneity of OCA conditions in EMU Carlos Vieira | Isabel Vieira
- 43: Labor Adjustment Dynamics: An Application of System GMM Pedro Esperança
- 44: Corporate taxes and the location of FDI in Europe using firm-level data Tomás Silva | Sergio Lagoa
- 45: Public Debt Stabilization: Redistributive Delays versus Preemptive Anticipations Paulo Júlio

46: Organizational Characteristics and Performance of Export Promotion Agencies: Portugal and Ireland compared Inês Ferreira | Aurora Teixeira

REPÚBLICA PORTUGUESA

ECONOMIA E MAR

- 47: Evaluating the forecast quality of GDP components: An application to G7 Paulo Júlio | Pedro Esperança
- 48: The influence of Doing Business' institutional variables in Foreign Direct Investment Andreia Olival
- 49: Regional and Sectoral Foreign Direct Investment in Portugal since Joining the EU: A Dynamic Portrait Irina Melo | Alexandra Lopes
- 50: Institutions and Firm Formation: an Empirical Analysis of Portuguese Municipalities Simão Arouca
- 51: Youth Unemployment in Southern Europe João Leão | Guida Nogueira
- 52: Financiamento da Economia Portuguesa: um Obstáculo ao Crescimento? João Leão | Ana Martins | João Gonçalves
- 53: O Acordo de Parceria Transatlântica entre a UE e os EUA constitui uma ameaça ou uma oportunidade para a Economia Portuguesa? João Leão | Guida Nogueira
- 54: Prescription Patterns of Pharmaceuticals Ana Gonçalves
- 55: Economic Growth and the High Skilled: the Role of Scale Eects and of Barriers to Entry into the High Tech Pedro Gil | Oscar Afonso | Paulo Brito
- 56: Finanças Públicas Portuguesas Sustentáveis no Estado Novo (1933-1974)? Ricardo Ferraz
- 57: What Determines Firm-level Export Capacity? Evidence from Portuguese firms Ana Gouveia | Ana Luisa Correia
- 58: The effect of developing countries' competition on regional labour markets in Portugal Tiago Pereira
- 59: Fiscal Multipliers in the 21st century Pedro Brinca | Hans Holter | Per Krusell | Laurence Malafry
- 60: Reallocation of Resources between Tradable and Non-Tradable Sectors in Portugal: Developing a new Identification Strategy for the Tradable Sector Ana Fontoura Gouveia | Filipa Canas
- 61: Is the ECB unconventional monetary policy effective? Inês Pereira
- 62: The Determinants of TFP Growth in the Portuguese Manufacturing Sector Daniel Gonçalves | Ana Martins





- 63: Practical contribution for the assessment and monitoring of product market competition in the Portuguese Economy – estimation of price cost margins Luis Folque
- 64: The impact of structural reforms of the judicial system: a survey Ana Gouveia | Silvia Santos | Corinna Herber
- 65: The short-term impact of structural reforms on productivity growth: beyond direct effects Ana Gouveia | Silvia Santos | Inês Gonçalves
- 66: Assessing the Competitiveness of the Portuguese Footwear Sector Fábio Batista | José Matos | Miguel Matos
- 67: The empirics of agglomeration economies: the link with productivity Ana Gouveia | Silvia Santos | Marli Fernandes
- 68: Determinants of the Portuguese GDP stagnation during the 2001-2014 period: an empirical investigation Carlos Figueira
- 69: Short-run effects of product markets' deregulation: a more productive, more efficient and more resilient economy? Ana Gouveia | Silvia Santos | Gustavo Monteiro
- 70: Portugal: a Paradox in Productivity Ricardo Pinheiro Alves
- 71: Infrastructure Investment, Labor Productivity, and International Competitiveness: The Case of Portugal Alfredo Pereira | Rui Pereira
- 72: Boom, Slump, Sudden stops, Recovery, and Policy Options. Portugal and the Euro Olivier Blanchard | Pedro Portugal
- 73: Case Study: DBRS Sovereign Rating of Portugal. Analysis of Rating Methodology and Rating Decisions Annika Luisa Hofmann | Miguel Ferreira | João Lampreia
- 74: For Whom the Bell Tolls: Road Safety Effects of Tolls on Uncongested SCUT Highways in Portugal Alfredo Pereira | Rui Pereira | João Pereira dos Santos
- 75: Is All Infrastructure Investment Created Equal? The Case of Portugal Alfredo Pereira | Rui Pereira
- 76: Why Virtuous Supply-Side Effects and Irrelevant Keynesian Effects are not Foregone Conclusions: What we Learn from an Industry-Level Analysis of Infrastructure Investments in Portugal Alfredo Pereira | Rui Pereira
- 77: The Role of Gravity Models in Estimating the Economic Impact of Brexit Graham Gudgin | Ken Coutts | Neil Gibson | Jordan Buchanan

- 78: Infrastructure Investment in Portugal and the Traded/Non-Traded Industry Mix Alfredo Pereira | Rui Pereira
- 79: Goods and Factor Market Integration: A Quantitative Assessment of the EU Enlargement Lorenzo Caliendo | Fernando Parro | Luca David Opromolla | Alessandro Sforza
- 80: Understanding productivity dynamics:a task taxonomy approach Tiago Fonseca | Francisco Lima | Sonia C. Pereira
- 81: On the Effects of Infrastructure Investments on Industrial CO2 Emissions in Portugal Alfredo Pereira | Rui Pereira
- 82: Assessing Competition With the Panzar-Rosse Model: An empirical analysis of European Union banking industry Suzana Cristina Silva Andrade
- 83: Health Care Investments and Economic Performance in Portugal: An Industry Level Analysis Alfredo Pereira | Rui Pereira | Pedro G. Rodrigues
- 84: Is deregulation of product and labour markets promoting employment and productivity? A difference-in-differences approach Hugo Correia | Ana Fontoura Gouveia
- 85: Foreign acquisition and internal organization Paulo Bastos | Natália P. Monteiro | Odd Rune Straume
- 86: Learning, Prices, and Firm Dynamics Paulo Bastos | Daniel A. Dias | Olga A. Timoshenko
- 87: The Diffusion of Knowledge via Managers' Mobility Giordano Mion | Luca David Opromolla | Alessandro Sforza
- 88: Empresas Zombie em Portugal Os sectores não transacionáveis da Construção e dos Serviços
 Gabriel Osório de Barros | Filipe Bento Caires | Dora Xarepe Pereira
- 89: Collective bargaining through the magnifying glass: A comparison between the Netherlands and Portugal Alexander Hijzen | Pedro Martins | Jante Parlevliet
- 90: A Lower VAT Rate on Electricity in Portugal: Towards a Cleaner Environment, Better Economic Performance, and Less Inequality Alfredo Pereira | Rui Manuel Pereira
- 91: Who Seeks Re-Election: Local Fiscal Restraints and Political Selection Susana Peralta | João Pereira dos Santos
- 92: Assessing the Competitiveness of the Metalworking Sector João Marinho | Pedro Carvalho





- 93: The efficiency of Portuguese Technology Transfer Offices and the importance of university characteristics Aurora Teixeira | André Monteiro
- 94: Persistence in innovation and innovative behavior in unstable environments Joana Costa | Anabela Botelho | Aurora Teixeira
- 95: The effect of entrepreneurial origin on firms' performance - The case of Portuguese academic spinoffs Natália Barbosa | Ana Paula Faria
- 96: Absorptive Capacity and Firms' Generation of Innovation - Revisiting Zahra and George's Model Dina Pereira | João Leitão
- 97: Innovations in digital government as business facilitators: implications for Portugal João Martins | Linda Veiga
- 98: Innovation and the economic downturn: Insights from Portuguese firms Hugo Pinto | Tiago Santos Pereira | Elvira Uyarra
- 99: European Funds and Firm Dynamics: Estimating Spillovers from Increased Access João Pereira dos Santos | José Tavares
- 100: Corporate Leverage and Investment in Portugal Ana Martins | José Henrique Gonçalves | João Mário Ferreira Duque
- 101: The effects of official and unofficial information on tax compliance Filomena Garcia | Luca David Opromolla | Andrea Vezzulli | Rafael Marques
- 102: Competition effect on innovation and productivity - The Portuguese case Anabela Santos | Michele Cincera | Paulo Neto | Maria Manuel Serrano
- 103: Measuring the Welfare of Intermediation in Vertical Markets Javier D. Donna | Pedro Pereira | Tiago Pires | Andre Trindade
- 104: Of course Collusion Should be Prosecuted. But Maybe... Or (The case for international antitrust agreements)
 Filomena Garcia | Jose Manuel Paz y Minõ | Gustavo Torrens
- 105: Product market competition and gender discrimination Dudley Cooke | Ana P. Fernandes | Priscila Ferreira
- 106: Integration of Small Technology-Based Firms in Aeronautics Anabela Reis | Joana Mendonça | Ligia Urbina
- 107: The Effects of Highway Tolls on Private Business Activity – Results from a Natural Experiment

João Pereira dos Santos | David B. Audretsch | Dirk Dohse

- 108: Competition and Firm Productivity: Evidence from Portugal Pedro Carvalho
- 109: Do Exchange Traded Funds (ETFs) Outperform the Market? Evidence from the Portuguese Stock Index Carlos Manuel Pinheiro | Hugo Hilário Varela
- 110: Assessing the Competitiveness of the Portuguese Chemical Sector Ana Rita Margues | Cátia Silva
- 111: A General Equilibrium Theory of Occupational Choice under Optimistic Beliefs about Entrepreneurial Ability Michele Dell'Era | Luca David Opromolla | Luis Santos-Pinto
- 112: O Mercado Segurador em Portugal: O Papel dos Gestores na Constituição de Provisões Soraia de Sousa Bornett | Carlos Manuel Pinheiro
- 113: Exploring the implications of di erent loan-tovalue macroprudential policy designs Rita Basto | Sandra Gomes | Diana Lima
- 114: The Determinants of TFP Growth in the Portuguese Service Sector Ana Martins | Tiago Domingues | Catarina Branco
- 115: Agglomeration and Industry Spillover Effects in the Aftermath of a Credit Shock José Jorge | Joana Rocha
- 116: Entrepreneurial Human Capital and Firm Dynamics Francisco Queiró
- 117: Global Value Chains and Vertical Specialization: The case of Portuguese Textiles and Shoes exports Tiago Domingues
- 118: Firm heterogeneity and exports in Portugal: Identifying export potential Frederico Oliveira Torres
- 119: Vantagens Comparativas Reveladas e suas determinantes: Uma Aplicação à Economia Portuguesa Guida Nogueira | António Portugal Duarte
- 120: A Look at the main channels of Potential Impact of Brexit on the Portuguese Economy Guida Nogueira | Paulo Inácio
- 121: How internationalization and competitiveness contribute to get public support to innovation? The Portuguese case Anabela Santos, Michele Cincera, Paulo Neto | Maria Manuel Serrano
- 122: Grande Guerra e Guerra Colonial: Quanto Custaram aos Cofres Portugueses? Ricardo Ferraz



- 123: Financing a Renewable Energy Feed-in Tariff with a Tax on Carbon Dioxide Emissions: A Dynamic Multi-Sector General Equilibrium Analysis for Portugal Rui M. Pereira | Alfredo M. Pereira
- 124: Brown Sugar, how come you taste so good? The impact of a soda tax on prices and consumption Judite Gonçalves | João Pereira dos Santos
- 125: ARFIMA Reference Forecasts for Worldwide CO2 Emissions and the National Dimension of the Policy Efforts to Meet IPCC Targets José Beirute | Alfredo M. Pereira
- 126: Reference Forecasts for CO2 Emissions from Fossil-Fuel Combustion and Cement Production in Portugal José M. Belbute | Alfredo M. Pereira
- 127: Regulated Early Closures of Coal-Fired Power Plants and Tougher Energy Taxation on Electricity Production: Synergy or Rivalry? Alfredo Marvão Pereira | Rui Manuel Pereira
- 128: Picking Our Environmental Battles: Removal of Harmful Subsidies or Carbon Taxation? Alfredo Marvão Pereira | Rui Marvão Pereira
- 129: Financing Future Feed-in Tariffs from Currently Installed RES-E Generating Capacity Alfredo Marvão Pereira | Rui Marvão Pereira
- 130: Foreign Direct Investment, Income Inequality and Poverty in Portugal, 1973-2014: What does cointegration analysis tell us? Aurora Teixeira | Ana Sofia Loureiro
- 131: On the Spillover Effects of CO2 Taxation on the Emissions of other Air Pollutants Alfredo Marvão Pereira | Rui Marvão Pereira
- 132: On the Macroeconomic and Distributional Effects of the Regulated Closure of Coal-Operated Power Plants Alfredo Marvão Pereira Rui Manuel Pereira
- 133: The China Shock and Employment in Portuguese Firms Lee Branstetter | Brian Kovak | Jacqueline Mauro | Ana Venâncio
- 134: Energy Taxation Reform with an Environmental Focus Alfredo Marvão Pereira | Rui Manuel Pereira
- 135: ARFIMA Reference Forecasts for Worldwide CO2 Emissions and the Need for Large and Frontloaded Decarbonization Policies José M. Belbute | Alfredo M. Pereira
- 136: Exporter Firms Behaviour, Evidence From Portuguese Firms Using Microdata Luís Pedro Manso Machado

- 137: Collateral Value and Entrepreneurship: Evidence from a Property Tax Reform Miguel Ferreira | João Pereira dos Santos | Ana Venâncio
- 138: The Financial Channels of Labor Rigidities: Evidence from Portugal Edoardo M. Acabbi | Ettore Panetti | Alessandro Sforza
- 139: Can a small leak sink a great ship? A comprehensive analysis of the Portuguese household savings Tiago Domingues | Margarida Castro Rego
- 140: Corporate taxes and high-quality entrepreneurship: evidence from a tax reform Ana Venâncio | Victor Barros | Clara Raposo
- 141: Built Like a House of Cards? Corporate Indebtedness and Productivity Growth in the Portuguese Construction Sector1 José Santos | Nuno Tavares | Gabriel Osório de Barros
- 142: Effectiveness of Simplex: The Case of Portuguese Social Security António Alberto Nifrário de Pinho Tavares
- 143: Digital innovation in higher education: A questionnaire to Portuguese universities and polytechnic institutes Paulo Nuno Vicente |Margarida Lucas | Vânia Carlos
- 144: Portugal in the Global Innovation Index: A panel data analysis Marcelo P. Duarte | Fernando M. P. O. Carvalho
- 145: Intangible investments and productivity performance Michele Cincera | Julie Delanote | Pierre Mohnen | Anabela Santos | Christoph Weiss
- 146: Digitalization in Two-sided Platform Competition Filomena Garcia | Muxin Li
- 147: Collusion between two-sided platforms Joana Pinho | Yassine Lefouili
- 148: Da confluência entre Big Data e Direito da Concorrência: As concentrações digitais - O caso Facebook/WhatsApp Ana Rodrigues Bidarra
- 149: The Determinants of Total Factor Productivity in the Portuguese Quaternary Sector Paulo Matos | Pedro Neves
- 150: Os modelos Input-Output, a estrutura setorial das economias e o impacto da crise da COVID 19





Pedro N. Ramos | João Ferreira | Luís Cruz | Eduardo Barata

- 151: Public Expenditure and private firm performance: using religious denominations for causal inference Henrique Alpalhão | Marta Lopes | João Santos| José Tavares
- 152: Employee Training and Firm Performance: Quasi-experimental evidence from the European Social Fund Pedro S. Martins
- 153: Dream Jobs Luca David Opromolla | Giordano Mion | Gianmarco I.P. Ottaviano
- 154: Minimum wage and financially distressed firms: another one bites the dust F. Alexandre | P. Bação | J. Cerejeira | H. Costa | M. Portela
- 155: Do short-term rentals increase housing prices? Quasi-experimental evidence from Lisbon Susana Peralta | João Pereira dos Santos | Duarte Gonçalves
- 156: Economic and social policies under EMU

Ricardo Pinheiro Alves

- 157: International Sourcing in Portuguese Companies - Evidence from Portuguese Micro Data Ana Martins | Guida Nogueira | Eva Pereira
- 158: The Impact of R&D tax incentives in Portugal Rita Bessone Basto | Ana Martins | Guida Nogueira
- 159: The Determinants of Competitiveness of the Portuguese Defense Industry | Roxanne Merenda
- 160: How is the Minimum Wage Shaping the Wage Distribution: Bite, Spillovers, and Wage Inequality Carlos Oliveira





